

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Rickne, Johanna

Working Paper

Gender, wages and social security in China's industrial sector

Working Paper, No. 2010:8

Provided in Cooperation with:

Department of Economics, Uppsala University

Suggested Citation: Rickne, Johanna (2010): Gender, wages and social security in China's industrial sector, Working Paper, No. 2010:8, Uppsala University, Department of Economics, Uppsala, https://nbn-resolving.de/urn:nbn:se:uu:diva-125153

This Version is available at: https://hdl.handle.net/10419/82536

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





Working Paper 2010:8 Department of Economics

Gender, Wages and Social Security in China's Industrial Sector

Johanna Rickne

Department of Economics Uppsala University P.O. Box 513 SE-751 20 Uppsala Sweden

Fax: +46 18 471 14 78

Working paper 2010:8 May 2010 ISSN 1653-6975

GENDER, WAGES AND SOCIAL SECURITY IN CHINA'S INDUSTRIAL SECTOR

JOHANNA RICKNE

Gender, Wages and Social Security in China's Industrial Sector

Johanna Rickne¹

Abstract

This study compares average earnings and productivities for men and women employed in roughly 200,000 Chinese industrial enterprises. Women's average wages lag behind men's wages by 11%, and this result is robust to the inclusion of non-wage income in the form of social insurance payments. The gender-wage gap is wider among workers with more than 12 years of education (28%), mainly because of the higher relative wages received by skilled men in foreign-invested firms. Women's average productivity falls behind men's productivity by a larger margin than the gap in earnings, and the null-hypothesis of earnings discrimination is thereby rejected. Equal average wages between men and women are found among firms located in China's Special Economic Zones, and also among some light industrial sectors with high shares of female employees. Market reform hence appears to have improved women's relative incomes.

Keywords: China, gender wage gap, non-wage compensation

JEL classification code: I30; J16; J71; O10

¹ Department of Economics, Uppsala University, P.O. Box 513, SE-75120, Uppsala, Sweden; Research Institute of Industrial Economics (IFN), P.O. Box 55665, SE-10215, Stockholm, Sweden; Email: <johanna.rickne@nek.uu.se>

1. Introduction

Investigating gender-earnings discrimination in China is an urgent issue. Not only is the question of great economic and social importance to the female half of the labor force, reaching 640 million in 2004. It is also of wider economic importance as discrimination discourages worker-groups from participating in employment and investing in skill development and education. As a consequence, human capital becomes underdeveloped and underutilized, and long-term growth is negatively affected. Before 1978, the Chinese industrial employee was assigned to a state-owned workplace where he or she received welfare and wages according to a pre-determined and egalitarian grading system. After thirty years of market reform, today's industrial workers can be hired and fired by firms under various forms of ownership. Once employed, they now receive work compensation that is largely determined by the firm. Researchers have argued that this dramatic increase in employer discretion when determining the employee's work compensation has led to a polarization of economic power across gender lines (Ngo, 2002; Razavi, 2007; Wang, 2006).

In this paper I assess the gender-wage gap in the Chinese industrial sector using firm-level data. The purpose is to examine how women fare compared to men in firms under various forms of ownership, and if the gender wage-gap differs between low- and high-educated employees. By testing if women's average wages correspond to their productivity contributions, I then test if women are subject to wage-discrimination. In an extended analysis, the effect of non-wage compensation on the gender-polarization of earnings is investigated by adding firm contributions to social insurances to the wage-measure.

The paper addresses a number of limitations in the existing literature, the first of which concerns the standard methodology. A vast majority of previous empirical studies of gender-wage discrimination in China uses individual- or household-level, where a residual gap in wages is interpreted as discrimination (see Shen and Deng, 2008 for a survey). As

pointed out by e.g. Altonji and Blank (1999), the observable individual-level characteristics in the wage-equation are however unlikely to capture all gender-related differences in productivity. Relying on this method therefore implies a risk of overestimating the degree of discrimination². By instead employing the method proposed by Hellerstein and Neumark (1999), this paper uses firm-level data to retrieve direct measures of the productivities of men and women from the production function. Discrimination can then be assessed by comparing the parameter capturing women's relative productivity with their relative wages.

A second contribution stems from the quality of the data used. In this respect, the paper provides substantial improvements on Dong and Zhang's (2009) unique firm-level study of gender-wage discrimination in China, a study that uses a methodology similar to the one in this paper. Compared to their study, data coverage is expanded from a fraction of the economy – 856 firms located in a few large cities – to a sample of 199,840 firms, including all large and medium-sized industrial enterprises as well as the majority of those designated as "small". This reduces the risk of sample bias³ and strengthens the statistical precision. Moreover, the availability of gender-disaggregated information on the education levels of firm employees allows the computation of skill-weighted wage- and productivity gaps between men and women.

This is the first study that examines the role of non-wage compensation to the gender-polarization of total earnings in Chinese industry. Providing employees with mainstream social insurances has been shown to carry a cost that amounts to over 50% of the total wage compensation paid by firms (Banister, 2005)⁴, and revious research has argued that

² The part of the wage-gap not explained by proxies for individual productivity is indeed large in previous studies, for example 53-63% found by Gustafsson and Li (2000), 89-92% by Liu et al (2000) and 85% by Wang and Cai (2008).

³ this is particularly important in a vast economy such like the Chinese where development levels, market reform intensity and social norms vary greatly across geographical regions.

⁴ Banister (2005) draws upon a survey conducted by China's Ministry of Labor, and covering 11,704 urban enterprises in 51 large cities, to conclude that the standard wage measure of employee earnings should be increased by 53.8 percent to fully account for labor compensation actually paid by urban employers.

insurance coverage has become increasingly gender-biased in the transition period (Razavi, 2007; Wang, 2006).

The results show an average skill-weighted wage-gap of 11% in the nation-wide sample, and a somewhat wider gap (28%) among men and women with above-high school education. Highly skilled men have benefitted more from the internationalization of the industrial sector. Their wage-advantages over skilled women are larger in firms with investments from abroad or from Hong Kong, Taiwan, or Macao. Another finding is that industrial sector matters greatly for the average male-female wage-gap. Zero wage-differences are found in sectors with large shares of female employees such as the Tailoring, Leather, and Textile sectors. Equal wages between women and men are also found among firms located in China's Special Economic Zones. Market reform and the increasing importance of light industry hence appear to have improved the relative incomes of female industrial employees.

Women's productivity lags behind men's productivity by a larger margin (41%) than their wages lag behind men's wages (11%). Although the null-hypothesis of wage-discrimination is thereby rejected, women's productivity disadvantage may reflect other forms of discriminatory treatment on the labor market (Bergmann, 1974)⁵. Some sectors have however been more successful in harnessing female productivity, especially foreign-invested enterprises and firms in light industrial sectors.

Non-wage compensation measured by firm contributions to social insurances is not found to increase men's earnings advantage. If anything, their distribution favors skilled workers in general and skilled women in particular, yielding a slightly narrower gender-wage gap among skilled workers than when only wages are considered.

The paper is organized as follows. Section 2 provides a brief overview of reforms to wage-setting and welfare systems in the post-1978 period and presents previous research on

4

⁵ Systematically lower productivities among women could result from gender-differences in firm's hiring practices and in occupational mobility, which in turn affect men's and women's efforts on their jobs.

gender-wage discrimination in China. Section 3 explains the basic econometric framework and section 4 discusses the dataset and variable computations. Estimation results are presented in section 5 and section 6 concludes.

2. Wage and non-wage compensation in Chinese industry

In pre-reform socialist China, urban industrial workers enjoyed a system of guaranteed occupational and income security. This "Iron Rice Bowl" also provided access to welfare benefits (health care, pensions, education) which were distributed via state-owned work units. The "emancipation of women" constituted an ethical commitment of the socialist state, and entailed their juridical equality with men, entry into paid work, and access to social rights. Under this system, women's incomes improved radically. The gender wage differential narrowed and became small in an international comparison, particularly in urban areas (Croll, 1995). With an industrial policy that emphasized heavy and capital-intensive industry, many women came to fill positions for which they were biologically disadvantaged compared to men, such as blue collar jobs requiring physical strength. The resulting skill mismatch was however not reflected in the centrally determined and egalitarian remuneration system (Korzec, 1992).

In the late 1970s, a series of radical reforms were enacted on the industrial labor market. Transformation of the wage-setting system started in the early 1980s when firms were given autonomy over their remuneration systems within government guidelines. By the 1990s, those guidelines had evolved into abiding by minimum wages (Shen, 2007). Another centerpiece of reform was the transfer of the labor allocation decision from the state and to the enterprises. By 1994, firms had been given the right to dismiss workers. This new authority was extensively practiced during a massive retrenchment program in the state-owned sector. Between 1997 and 2002, more than 28 million state employees were asked to leave their jobs

(Dong and Xu, 2005)⁶.

Reform aimed at streamlining SOE organizations came as a response to the increasing competition from the private sector, which was growing rapidly. This private sector, composed by both domestic and foreign-invested actors, also represented a problem from a social insurance perspective. A growing number of employees working in the private firms did not have access to the enterprise-provided benefits granted by the State-owned work units. In fact, as market-reform progressed, even state employees were increasingly facing social insecurity. Many SOEs were having difficulties meeting their welfare obligations, partly because of the rising competition from the private sector, and partly because of a massive welfare burden stemming from a system setup where each SOE retained substantial financial responsibilities for its laid-off and retired workers. To free the SOEs from this welfare burden and to provide social security for private sector workers became two main objectives for a social system overhaul in the 1990s.

Put simply, welfare reform consisted in breaking up the enterprise-based entitlement model, known as the Labor Insurance Scheme (LIS), into separate social insurances. Instead of the self-insurance of each firm under the LIS, the new programs aimed to share risk across firms by pooling worker accounts on the provincial- and, in some cases, the city-level. State-owned firms were encouraged to switch from the LIS to the separate insurances, and private firms were mandated to join. In this way, the 90s saw the birth of a two-tiered pension system in 1997 and the Urban Employee's Basic Health Care Insurance System in 1998. In 1999, China instituted an unemployment insurance system with properties similar to those found in industrialized countries⁷. Concerning the provision of housing, the direct allocation by work

⁶ Women were over-represented among the laid-off SOE employees (Appleton et al., 2002; Giles et al., 2006) but simultaneously increased in demand in the export-oriented, and highly labor intensive, industries.

⁷ Under this program, an individual is eligible for benefits if he or she i) has contributed to the Unemployment Insurance for at least one year, ii) has been involuntarily laid-off, and iii) is registered as unemployed and willing to work. The most recent alteration in to the Unemployment Insurance system is the *binggui* policy under which previously laid-off SOE workers are detached from their old workplaces, where they had remained despite being formally out of work and merged with the pooled insurance system.

units visa the LIS was replaced by a savings system where accumulated funds can be used for housing purchase or repairs⁸. As with the LIS, these new insurances applied to urban industrial firms, while rural industry was excluded, although with the ambition of a later extension of coverage. Rural areas are also excluded from formal medical insurance, and retirement protection is scarce (Banister, 2005; Jackson et al., 2009).

The new insurance programs had to make up for a legacy of unmet obligations in the SOE sector. Although subject to some discretion at the administrative level (provincial and sometimes sub-provincial), contribution levels were therefore set at a generally high level. Guidelines call for a total of 24% of payroll directed toward pensions accounts (Jackson et al, 2009), 16% toward the housing accumulation funds (by year 2003, Wang et al., 2005), between 2-6% toward medical insurance (Xu et al, 2007), and 2% toward the unemployment insurance (Vodopivec and Tong, 2008).

Similar to the ambition for mobile accounts in the new insurance systems, the ambition of complete coverage of the urban industrial sector remains largely unmet. Although participation in the programs is mandatory for firms, lack of enforcement has effectively given them the option of non-compliance. In 2005, the pension scheme covered 17% of the workforce, and the unemployment insurance covered 14% (NBS, National Bureau of Statistics of China, 2006, pp. 43 and 201). Overall health insurance coverage actually decreased between 1998 and 2003, as mainstream coverage fell more sharply than the increase in commercial and other non-commercial insurances (Xu et al., 2007). Qualitative research has indicated an emerging pattern of gender-polarization in insurance coverage in the export-oriented sectors of the economy (Razavi, 2007), and that in particular women dismissed from SOEs have been pushed into ownership sectors where social security protection is low (Cooke, 2001; Stockman, 1994). It has also been pointed out that the

⁸ In addition to participating in these housing accumulation funds, firms may also provide their employees with loosely regulated subsidies. These may cover a range of residential services, rent payments as well as cash hand outs to purchase state-owned housing which was previously allocated by the work units (Wang et al., 2005).

principle of proportionally tying contributions to the employee's wage may compound the wage-differential between men and women (Razavi, 2007; Wang, 2006). These claims however remain without quantitative verification.

Contrary to the case of non-wage compensation, a considerably large literature has investigated the wage-aspect of earnings discrimination in the reform period. Most previous studies use individual-level data from the China Household Income Project, carried out by the Chinese Academy of social sciences in 1988 and 1995. They find that the gender wage gap is slightly below 20%, and fairly constant over this period (Gustafsson and Li, 2000, Bishop et al. 2005, Démurger et al., 2007, Shu and Bian, 2003). This result is corroborated by earlier findings from a Tianjin sample collected in 1993 (Bian et al., 2000), as well as the findings of researchers using data from the Urban Household Survey between 1988 and 1999 (Ng, 2004; Liu et al, 2004). Xu et al. (2006) find a slightly larger gap, 32%, in two recently urbanized towns in Zhejiang Province between 1999 and 2000. This is similar to Dong and Zhang's (2009) finding when using firm-level data from the late 1990s. Regarding the size of the wage-gap across educational categories, a number of studies find that the gap is narrower among highly educated employees (Xu et al., 2006; Hughes and Maurer-Fazio, 2002; Gustafsson and Li, 2000).

The share of the gender-wage gap that is not explained by gender-related productivity proxies in the wage equation is generally large in previous studies. Over 50% of the wage gap is found to be unexplained by some studies (Bishop et al. 2005, Shu and Bian, 2003, Gustafsson and Li, 2000), and over 75% by many others (Liu et al. 2000; Rozelle et al. 2002; Wang and Cai, 2008). On the contrary, Dong and Zhang's (2009) study does not find evidence of discrimination. Women's estimated wages are not statistically different from their estimated productivity.

In a transition economy, the wage-setting practices of employers may differ widely

between the large ownership fractions on the labor market. Xu et al. (2006) record the largest wage gap in privately owned enterprises, while Maurer-Fazio and Hughes (2002) and Maurer-Fazio et al. (1999) find the largest gaps in firms registered as joint ventures. Liu et al. (2000), and Hughes and Maurer-Fazio (2002), also find the widest wage-gaps in the most marketized ownership sectors. Their results however also show that these sectors have the largest shares of their wage-gaps remaining unexplained by observed worker characteristics. Rozelle et al. (2003) do not find any systematic association between the level of wage discrimination and the degree of market orientation by industry or ownership. Finally, the firm-level study of 856 firms conducted by Dong and Zhang (2009) yields the result that women are rewarded in accordance with their productivity in private firms, but that women are over-compensated in the state-owned sector.

3. Empirical methodology

We begin by expressing firm value added as a Cobb Douglas production function with the inputs capital (K) and labor (L) ⁹. The labor input is interpreted as a quality of labor index

$$QL = M_{U} + \phi_{FU} F_{U} + \phi_{MS} M_{S} + \phi_{FS} F_{S}$$
 (1)

where F_j and M_j are the number of female or male employees who are skilled, j=S, or unskilled, j=U. In logs, our production function can be written

$$\ln(VA) = \ln(A) + \alpha_C \ln(K) + \alpha_L \ln[M_U + \phi_{FU} F_U + \phi_{MS} M_S + \phi_{FS} F_S]. \tag{2}$$

-

 $^{^{9}}$ As discussed by Griliches and Ringstad (1971), the value added specification of the production function improves comparability of data across industries and across establishments within industries. It also allows greater comparability when industries or establishments differ in the degree of vertical integration, and can be derived from quite polar production function specifications: one in which the elasticity of substitution between materials and value added is infinite (i.e., Y=f(K,QL)+M); and one in which this elasticity of substitution is zero (so that materials have to be used in a fixed proportion to output).

The specification of the quality of labor index implies that members of the four gender-skill groups are assumed to be perfectly substitutable inputs, but that the marginal productivities of the groups are allowed to differ. The parameters ϕ_{ij} denote the average marginal productivity of employees with male or female gender i = (M, F) that are considered to be either skilled or unskilled j = (S, U) as compared to male unskilled workers. In this setting, the empirical observation that $\phi_{FU} = \phi_{MS} = \phi_{FS} = 1$ would indicate that all four gender-skill groups contribute equally to firm output. Alternatively, the interpretation of a finding that $\phi_{FU} = 0.5$ would be that the average marginal productivity of female unskilled employees is 50% lower than the average productivity of their male unskilled co-workers. Letting L represent total firm employment so that $M_U = L - F_U - M_S - F_S$ we can rewrite (2) as

$$\ln(VA) = \ln(A) + \alpha_C \ln(K) + \alpha_L \ln[L + (\phi_{FU} - 1)F_U + (\phi_{MS} - 1)M_S + (\phi_{FS} - 1)F_S].$$
 (3)

We next turn to the estimation of relative differentials in work compensation. A firm-level wage equation is set up as a Mincer-type earnings equation

$$\ln(W_T) = \lambda_0 + \ln(QL) = \lambda_0 + \ln[L + (\lambda_{FU} - 1)F_U + (\lambda_{MS} - 1)M_S + (\lambda_{FS} - 1)F]$$
 (4)

where the dependent variable is the total wage bill, or the total amount of worker compensation, paid by the firm¹⁰. Analogous to the production function specification, the

¹⁰ To see how this firm-level function can be understood as an aggregation of individual-level wage equations over workers in the firm, consider the individual level wage equation $w_i = w_M M_i + w_F F_i$ where w_i is the wage of an individual worker with gender i, w_M and w_F are average wages, and M_i and F_i are dummy variables for females and males respectively. Aggregating this function over the firm we get that the total wage bill is $W = w_M (L - F) + w_F F$, which can be expressed as $W = w_M \left[L + (\lambda_F - 1)F \right]$ where λ_F is the average relative wages of women compared to men, $\frac{w_F}{w_M}$. Taking logs gives

parameters λ_{ij} denote the relative average wages of the three gender-skill groups compared to the wages of unskilled male workers.

By comparing the estimated results of the production function (3), and the wage equation (4) we can now specify a test of whether or not gender-skill groups of employees are paid in accordance with their contribution to firm output. This test amounts to comparing the relative wage parameters λ_{ij} with the relative productivity parameters ϕ_{ij} for each group. If the estimated marginal work compensation falls short of the marginal productivity so that $\lambda_{ij} < \phi_{ij}$, this is interpreted as under-compensation of that specific gender-skill group. To produce the parameters necessary to test whether women, as an aggregated group, are discriminated against, we compute skill-weighted productivity and earnings differentials as

$$\phi_F = \frac{\phi_{FU} P_{FU} + \phi_{FS} P_{FS}}{P_{MU} + \phi_{MS} P_{MS}} \tag{5}$$

where P_{ij} is the proportion of employees with gender i and skill level j in their respective gender group 11 , where ϕ_{ij} is replaced by λ_{ij} to produce the earnings differential.

A major limitation of using firm-level data is the inability of distinguishing whether estimated differences in wages and productivities between worker groups stem from within or across-firm variation. The interpretation of a negative gap between average wage and productivity of a socioeconomic group ($\lambda_{ij} - \phi_{ij} < 0$), as discrimination, hence hinges on the assumption that this group has not been systematically hired by lower paying or less productive firms. Regarding unobservables, their effect on the test of equal relative productivities and earnings should be limited if the direction of the bias is the same for both

the simplified equivalent of equation (4), in which the constant corresponds to the average wage of men $\lambda_0 = \ln w_M$.

 $^{^{11}}$ so that $P_{FU}+P_{FS}=1\,,$ and $P_{\scriptscriptstyle MU}+P_{\scriptscriptstyle MS}=1\,\cdot$

variables. Robustness checks carried out below aim to assess potential bias in the point estimates stemming from, among other thing, gender-differences in the number of hours worked, selection of women into labor-intensive industries, and the degree of market exposure facing the firm.

Estimation of equations (3) and (4) is done simultaneously using the Non-Linear Seemingly Unrelated Regressions (NLSUR) method which takes account of cross-equation correlation in the shocks to wages and output. A vector of controls is added to each of the estimated production and wage equations. The measures of work compensation and output of firms are thereby allowed to vary systematically with the firms' age, size, geographical location and industrial sector¹².

4. Data, variables and summary statistics

This study uses survey data on industrial firms collected by China's National Bureau of Statistics (NBS) in 2005¹³. The dataset covers all state-owned firms and all non-state firms with annual sales above 5 million RMB (about 750,000 US dollar) and includes all firms formally designated as large or medium size, as well as the bulk of those designated as "small-scale." Dividing the total value added in the dataset by the industrial GDP reported in China's Statistical Yearbook indicates that the dataset accounts for 94% of total industrial output.

The dataset contains information on sales, capital stock, the total number of workers and the share of women, the total wage bill, and also annual firm expenditures on four categories of social insurance items: (1) housing accumulation funds and housing subsidies,

reports in the China Markets Yearbook.

_

Adding these controls corresponds to assuming that firms are price takers in labor market specific to these characteristics (i.e. geographical location, industry etc). Moreover, as long as segmentation of markets along these dimensions leads to proportional variation in wages across gender, then the equations can be augmented by adding these variables linearly.
This dataset is used by the NBS to compile the "industry" section of the China Statistical Yearbook and industry specific

(2) pensions and medical insurance, and (3) labor insurance and unemployment insurance. In order to attain a more accurate measure of labor quality, the 2005 dataset is complemented with information on the gender-disaggregated human capital distribution of firms from the expanded census data from 2004. A worker is considered to be skilled if he or she has completed junior college or more, and unskilled otherwise. Merging data for these two years requires the assumption that the share of skilled persons within the female and male employee groups has not changed between the years. It also necessitates the removal of 61,384 firms which are not in operation in both years. The final sample size after following the data cleaning procedure of Jefferson et al. (2008) is 199,840¹⁴.

Variables are created as follows. The dependent variable of the production function is value added, which is calculated by subtracting material costs from total sales. The capital measure is the net value of fixed assets. Wages is the total wage bill, which includes social insurance payments deducted from employee earnings, but not the payments made by the employer (Table A1 lists the components of the wage variable). A more comprehensive measure of total work compensation is created by adding the sum of the employer's expenditures on social insurance payments to the wage bill. R&D intensity is the logarithm of the firm's expenditure on research and development, and a unionization dummy takes the value one if the firm the firm has a trade union in the year 2004, and zero otherwise.

Ownership categories are created by aggregating 23 ownership registration types into five broader categories, closely tracking the formal classification system currently used for reporting data on all industrial firms in China's Statistical Yearbook. Enterprises are defined as State-owned (SOE), Collective owned (COE)¹⁵, Private (PRI), funded by entities in Hong

¹⁴ Enterprises with less than eight employees, and those that are in the upper and lower tails of productivity, are excluded to correct for improbable values. The latter is done in two steps. First, ratios of value added to labor and capital, and ratios of labor and capital to value added are computed. Then, those firms that lay more than four standard deviations above the mean of each of those four variables are deleted. Firms are also dropped if the ratio of value added to sales is either negative or larger than one, and if they report negative values of capital or labor. The procedure removes 17,868 firms, whereof the majority report zero employment.

¹⁵ Collective-owned firms are economic entities that are registered in accordance with the Regulation of the People's

Kong, Macao or Taiwan (HKMT), Foreign-invested (FIE), and "Other" (see Table A2 for details). Also included in the empirical analysis are two measures of the asset ownership composition of firms: the shares of paid-in capital supplied by the state and the combined shares for capital from entities in foreign countries or from HKMT. Incorporating these measures may yield improved predictions of firm performance, in particular if ownership registration is an unreliable measure of actual firm control (Sabin, 1994; Jefferson and Su, 2006). Additional control variables are industrial categories on the two-digit level, province, size, age, and a proxy for rural location taking the value one if a firm is under the administrative control of the county, small town, street, village, resident- or village committee.

Summary statistics are reported in Table 1. They show that the average firm has 373 employees, among which women make up 42%. Of these women, 10.6% have a junior college degree or higher (>12 years of education). The corresponding percentage for men is 15.5%, illustrating the gender division in human capital in the industrial sector as evidenced in the national census conducted in 2000¹⁶. Examining the ownership-based sub-samples, the share of female employees is the lowest in the state-owned sector (29%), somewhat higher in firms that are private (43%) or collective owned (40%), and the highest in foreign- (52%) and HKMT-funded firms (54%). The state-owned firms have a larger average size (803) than the other sectors, and we find the smallest average workforce size in the collective and private sectors, close to 150 workers. Average wages are the lowest in the state-owned and collective owned enterprises.

R

Republic of China on the Management of Registration of Legal Enterprises, where assets are owned collectively. They include urban and rural enterprises invested by collectives, and some enterprises registered with industrial and commercial administration agency as collective units, where funds are pulled together by individuals who voluntarily give up their right of ownership (China Statistical Yearbook 2006, chapter 13).

¹⁶ In urban China, men had on average 0.93 more years of schooling than women.

¹⁷ Previous studies suggest that the workforces of export-oriented and foreign-invested enterprises may be up to eighty percent female (Tan, 2000). In the dataset used in this paper this is true for 12% of the HKMT-funded firms, and for 14% of the foreign-funded firms.

The ownership distribution of the dataset is such that 44.4% of the firms are domestic and privately owned, 7.9% are state-owned, 8.4% are collective owned, while 11.2% and 11.6% are funded by entities based in Hong Kong, Macao or Taiwan and in foreign countries respectively. Finally, 16.4% of the firms fall into the category "Other" Examining the average asset shares held by the Chinese state and by foreign and HKMT entities in the different ownership categories does not provide evidence for sizeable deviations between the registration-based ownership categorization and the asset holding structure of firms. Summary statistics of the geographical and industrial composition of the dataset are placed in Appendix, Tables A3 and A4.

Table 2 displays descriptive statistics of the three non-wage compensation expenditure items. The top half of the table shows the share of firms reporting positive expenditures on each variable, which is interpreted as participation in the program, and the bottom half details the cost of the insurance item as a share of the total wage cost in those participating firms. We first note that 16% of the firms report non-zero expenditures on all three non-salary items. Looking at the disaggregated figures, the highest participation rate is recorded for the Pensions and Medical care programs (66%), while a lowest is associated with the Labor and/or Unemployment insurances (44%) and the Housing funds and subsidies programs (18%). There is substantial variation in participation rates across ownership sectors. Coverage is higher than average among state-owned firms and foreign-invested firm¹⁹.

Turning to the bottom half of Table 2, we see that payments amount to, on average, 39% of the total wage cost in firms that report non-zero expenditures on all three items. For the housing variable, and the combined expenditures on health and pension insurance, we may moreover note that expenditure levels as a share of wages is markedly below the payment guidelines set up by the central government.

¹⁸ See Table A2 for a list of the registration categories included under this definition.

¹⁹ Because of the presence of rural firms with lower welfare obligations in the sample, the figures in Table 2 should not be interpreted as the precise share of urban firms that comply with social insurance requirements.

5. Are women discriminated against in Chinese industry?

5.1 Baseline estimation results

The first main results of this paper are found in Table 3. Columns 1 and 2 present estimates from jointly estimating equations (3) and (4) for the full sample and using the total wage bill as the dependent variable in the wage equation. We first note that that the estimated coefficients on capital and labor in the production function are 0.62 and 0.22, both of which are highly significant and of plausible magnitudes. Row one of the table contains the skill-weighted productivity and wage differentials between women and men. These parameters indicate that women earn 11% less than men ($\lambda_F = 0.89$) and that they are 41% less productive ($\phi_F = 0.59$). Comparing the sizes of the estimates to the results found by authors using similar empirical strategies for other economies, the narrow wage gap is particularly noteworthy. Hellerstein et al. (1999) report a 45% gender-wage difference for the US, Bartolucci (2008) finds a 34% gap in the case of Germany, and Asano and Kawaguchi (2007) a 70% gap in Japan. The comparatively narrow wage gap in China indicates that Maoist gender equality ideology has been maintained in the economic transition. Regarding the productivity

Table 1 Summary Statistics

	All firms	SOE	COE	PRI	HKMT	FIE	Other
No. of firms	199,840	15,788	16,821	88,791	22,373	23,282	32,785
(share of total sample)	(100)	(7.9)	(8.4)	(44.4)	(11.2)	(11.6)	(16.4)
Log value added	8.96	9.08	8.72	8.65	9.24	9.59	9.20
Log capital	8.40	9.43	7.95	7.93	8.69	8.99	8.79
Log wages	7.33	7.84	7.04	6.93	7.88	7.98	7.51
Log wages and benefits	7.44	8.08	7.14	7.01	7.97	8.10	7.64
Employment	283	806	185	146	370	361	340
Female employees (share)	0.42	0.29	0.40	0.43	0.54	0.52	0.38
Share of male unskilled employees	0.49	0.55	0.56	0.52	0.41	0.40	0.51
Share of male skilled employees	0.09	0.15	0.05	0.05	0.05	0.08	0.11
Share of female unskilled employees	0.38	0.23	0.39	0.41	0.51	0.47	0.33
Share of female skilled employees	0.05	0.07	0.02	0.03	0.03	0.05	0.05
Log R&D expenditure	0.58	1.05	0.26	0.38	0.48	0.74	1.00
Firm size (share)							
8-50	0.22	0.20	0.26	0.28	0.11	0.15	0.17
51-100	0.25	0.17	0.27	0.30	0.19	0.20	0.23
101-500	0.43	0.40	0.41	0.39	0.53	0.50	0.46
501-1000	0.06	0.11	0.04	0.03	0.11	0.09	0.08
1001-	0.04	0.13	0.02	0.01	0.07	0.07	0.06
Firm age (share)							
< 3 years	0.26	0.10	0.12	0.31	0.24	0.27	0.27
4-7 years	0.26	0.15	0.25	0.23	0.28	0.31	0.24
8-12 years	0.24	0.68	0.56	0.13	0.25	0.15	0.22
> 12 years	0.24	0.39	0.68	0.04	0.11	0.13	0.36
Rural administrative subordination	0.19	0.07	0.08	0.33	0.24	0.27	0.27
FIE and HKMT asset share	0.18	0.01	0.01	0.01	0.77	0.74	0.01
State asset share	0.07	0.81	0.01	0.00	0.01	0.01	0.01
Unionization	0.47	0.85	0.53	0.37	0.14	0.44	0.57

Notes: Value added, capital, wages, non-wage compensation, and R&D expenditures are in logs of thousands of Yuan.

Table 2 Coverage and economic importance of enterprise-based social insurance programs

	All three non-salary items	Pensions and health insurance	Housing funds	Labor and unemployment insurance					
Share of participating firms (expenditure > 0)									
	0.10	0.44							
All firms	0.16	0.66	0.18	0.44					
SOE	0.45	0.77	0.52	0.67					
COE	0.11	0.60	0.13	0.36					
PRI	0.06	0.56	0.34	0.32					
HKMT	0.13	0.81	0.15	0.58					
FIE	0.30	0.80	0.34	0.61					
Other	0.21	0.67	0.24	0.50					
Ratio of insurance payments to total wage costs in participating firms									
All firms	0.39	0.14	0.11	0.06					
SOE	0.44	0.24	0.11	0.09					
COE	0.37	0.16	0.09	0.06					
PRI	0.42	0.11	0.13	0.06					
HKMT	0.28	0.09	0.08	0.03					
FIE	0.29	0.13	0.08	0.03					
Other	0.37	0.17	0.13	0.06					

difference, it is larger than those found for the non-Asian economies (16% for both the US and Germany), but smaller than the one for Japan (66%). Testing the hypothesis of no discrimination by comparing the size of the two gaps shows that women in Chinese industry on average receive wages that exceed their productivity contributions. We hence reject the null-hypothesis of wage-discrimination against female employees in the nation-wide sample.

A productivity disadvantage of women is a common result in empirical studies that use firm-level data to estimate gender-differences in productivity from production functions. It should be noted that this disadvantage could in itself reflect gender-related discrimination in the allocation of labor within and between firms (i.e. Bergmann, 1974). A systematic difference in productivity between men and women with similar education levels may reflect that women are systematically matched with jobs that do not fit their preferences, skills and abilities. Previous studies of the Chinese labor market have found discriminatory treatment of women in the hiring process (Cooke, 2001), and in upward mobility within and between firms (Cao and Hu, 2007; Song and Dong, 2009). Such gender-stereotyping and discrimination may

in turn have affected Chinese women's motivations and efforts on their jobs (Korabik, 1994; Peng et al. 2009) and thereby led to further widening of the productivity gap.

We now compare the relative marginal wages and productivities of men and women with similar skill levels. Row three in Table 3 shows estimation results for workers with less than 12 years of education. Because unskilled workers make up the majority of the workforce in the industrial sector, these coefficients are very similar those in row one: women are 45% less productive and earn 11% less. Results for workers with more than 12 years of education are shown in rows 3 and 4. For both women and men, their wages and productivities are substantially larger than for unskilled males, as indicated by the parameters exceeding one. Investing in education is rewarded by a 93% education premium for men ($\lambda_{MS} = 1.93$), and a 65% (λ_{ES} =1.65) premium for women. Calculating the difference between these two premiums returns a 28% gender-wage gap among skilled workers (1.93-1.63). The divergence of this gender-wage gap from that found among unskilled workers (11%) is statistically different at the one percent level. Contrary to the results from some previous studies using individual data (Millimet and Wang, 2006; Hughes and Maurer-Fazio, 2002; Xu et al., 2006; Gustafsson and Li, 2000), the empirical analysis in this paper therefore indicates that the gender-wage gap is narrower in the lower part of the skill-distribution than in the upper part.

Next, we test the null of matching productivities and wages of skilled male and female workers. The results in rows seven and eight are both negative, suggesting that the wage premiums received by persons of both genders who invest in education fail to reflect the value of those investments to firm productivity. This result echoes previous findings

²⁰ Within the groups "skilled" and "unskilled", men and women have roughly the same average number of years of schooling.

regarding inefficiencies in the compensation of skilled employees in China's rural industry (see Fleisher and Wang, 2004)²¹.

Table 3
Joint production function and earnings equation estimates:
Cobb-Douglas production function, all firms

Coop-Douglas production function	<u> </u>		Log	Log
	Log (Value	Log	(Value	(Total
Wages and productivities relative to	Added)	(Wages)	Added)	Comp.)
unskilled male workers for	(1)	(2)	(3)	(4)
	0.59	0.89	0.59	0.90
female workers	(.01)	(.01)	(.01)	(.01)
	0.55	0.89	0.55	0.89
female unskilled workers	(.01)	(.01)	(.01)	(.01)
	2.43	1.65	2.42	1.82
female skilled workers	(.11)	(.03)	(.10)	(.03)
	3.92	1.93	3.88	2.01
male skilled workers	(.11)	(.02)	(.11)	(.02)
Wage-productivity gap for			-	
female workers	0.31		0.31	
female unskilled workers	0.34		0.34	
female skilled workers	-0.80		-0.60	
male skilled workers	-1.99		-1.87	
	0.62		0.63	
Log labor	(.01)		(.00)	
	0.22		0.21	
Log capital	(.00.)		(.00)	
	0.05	0.03	-0.06	-0.15
Log R&D expenditure	(00.)	(.00)	(.01)	(.00.)
	-0.06	0.12	-0.22	-0.05
Rural administrative subordination	(.01)	(.00.)	(.01)	(.01)
R-squared	0.56	0.84	0.56	0.83
Correlation between equations	0.13		0.13	
N	194,942		194,950	

Notes: All equations include a constant term and dummy variables for geographical location (22 provinces, 5 administrative regions and 4 municipalities, Beijing = reference), industry (38 categories, textile = reference), size (five categories, < 51 employees = reference), age (four categories, established earlier than 1994 = reference), unionization (1 if the firm has a union), ownership (six categories, private Chinese firms = reference), the share of state capital in total capital, and the share of foreign and HKMT capital in total capital. Bold letters indicate that the null-hypothesis of equal wages and productivities for workers of gender i and skill-group j, $\lambda_i - \phi_i = 0$ or $\lambda_{ij} - \phi_{ij} = 0$, is rejected at the five percent level. Standard errors are reported in parenthesis.

²¹ The authors find that monopsony power and inadequate possibilities to exploit returns to scale among firms do not satisfactorily explain the failure of wages of skilled workers to reflect their productivity. Persistant deviations of wages from productivity, they argue, could be the result of segmentation of the labor market caused the hukou-system's restrictions on worker mobility.

5.2 Does ownership matter?

Many studies of gender-wage discrimination in China have confirmed the importance of firm ownership. Table 4 presents results from jointly estimating equations (3) and (4) for six ownership sectors: 1) state-owned enterprises (SOE), 2) collective owned enterprises (COE), 3) privately owned enterprises (PRI), 4) firms funded by entities based in Hong Kong, Macao or Taiwan (HKMT), 5) foreign-invested firms (FIE), and 6) firms with mixed ownership (Other). We first focus on the gap in skill-weighted average wages between men and women, reported in row 1. We see that this gap is somewhat smaller in COEs (3%) and PRI (10%) firms than in FIEs (16%). The estimated gap among State-owned (12%) firms is however not statistically significant from any of the other ownership sectors. If we consider Collective firms as "state-controlled", these results are somewhat in line with previous studies showing larger wage-gaps in foreign-invested sectors than in state-controlled ones (Maurer-Fazio et al. 1999; Hughes and Maurer-Fazio, 2002; Liu et al., 2000; Xu et al., 2006).

By examining the gender wage-gaps between workers with similar education levels we can gain further insights into the importance of ownership sector to the wage dynamics between women and men. Looking first at the reward pattern between women and men with less than 12 years of education, and who comprise on average 87% of the firms' workforces, we note that unskilled women earn less than unskilled men in all sectors. In contrast to the combined skill-weighted measure of the total gender-wage gap discussed above, the point-estimates of the gap between unskilled men and women is the smallest in foreign-invested firms (5%) and the largest in state-owned enterprises (18%).

Next, we calculate the wage-gap among men and women with more than 12 years of education. This is done by subtracting the coefficient which captures the education premium of women compared to unskilled males (λ_{FS}) from the coefficient of the male education premium (λ_{MS}), reported in rows 3 and 4 respectively. For all ownership sectors, these

relative wage- parameters exceed zero, meaning that that human capital investments of both women and men are reflected in higher wages.

Market reform appears to have favored the wage growth of skilled men. The genderwage gap between skilled men and women is not statistically different form zero among SOEs, and skilled women are found to earn more than skilled men (by 59%) in COEs. In contrast, there is a significantly larger wage advantage of men in foreign-invested firms (117%) and in firms with investments from HKMT (75%). These results indicate that skilled men have been able to take advantage of the internationalization of ownership in the industrial sector to larger extent than skilled women.

Row 5 in Table 4 tests the hypotheses of equal wages and productivities of men and women by ownership sector. The results do not alter our conclusions from the overall firm sample. Women's productivities fall further behind men's productivities than their wages do. Put differently, women appear to receive wages that exceed their productivity in all the six ownership sectors under consideration. The largest disparity is found among collective owned enterprises and the smallest in workplaces funded by foreign entities or entities in Hong Kong, Macao, or Taiwan. Driving this result are the smaller productivity lags of women within the ownership sectors enjoying non-mainland investment. As such, the results suggest that female labor has been more efficiently allocated in the most internationalized sectors.

Increasing the share of skilled employees of both genders is associated with productivity gains for firms in all sectors. Rows 7 and 8 in Table 4 show the difference between the size of these productivity gains for the firms, and the size of the education premiums received by the employees. Echoing the result from the overall sample, we see that wage premiums fall short of reflecting the productivity contributions in all ownership sectors.

Table 4
Joint production function and wage equation estimates: Cobb-Douglas production function, ownership-based sub-samples

Joint production function and wage equation estimates. Coop-Douglas production function, ownersing-based sub-samples												
	SOE		CC	E	PRI HKMT		FIE		Other			
Wages and productivities relative to unskilled male workers for	Log (Value Added)	Log (Wage)										
	0.57	0.88	0.54	0.97	0.53	0.90	0.65	0.87	0.72	0.84	0.57	0.93
female workers	(.06)	(.03)	(.06)	(.02)	(.02)	(.01)	(.03)	(.01)	(.04)	(.02)	(.04)	(.02)
	0.70	0.82	0.41	0.92	0.48	0.91	0.69	0.92	0.72	0.95	0.51	0.91
female unskilled workers	(.08)	(.06)	(.05)	(.02)	(.02)	(.01)	(.04)	(.02)	(.04)	(.02)	(.04)	(.02)
	2.25	1.81	3.13	1.95	2.00	1.27	2.62	1.67	3.87	2.01	2.24	1.61
female skilled workers	(.37)	(.11)	(.62)	(.13)	(.17)	(.04)	(.27)	(.08)	(.32)	(.08)	(.25)	(.07)
	4.97	1.86	3.14	1.36	2.95	1.46	4.70	2.42	5.93	3.18	3.64	1.79
male skilled workers	(.43)	(.08)	(.47)	(.08)	(.15)	(.03)	(.35)	(.08)	(.43)	(.09)	(.24)	(.05)
Wage-productivity gap for												
female workers	0.31		0.43		0.37		0.22		0.12		0.36	
female unskilled workers	0.12		0.50		0.42		0.23		0.23		0.40	
female skilled workers	-0.44		-1.17		-0.73		-0.96		-1.86		-0.44	
male skilled workers	-3.11		-1.78		-1.49		-2.30		-2.75		-1.85	
	0.20		0.20		0.21		0.70		0.22		0.22	
Log capital	(.01)		(.01)		(.00)		(.02)		(.00.)		(.00.)	
	0.77		0.49		0.52		0.19		0.70		0.65	
Log labor	(.02)		(.02)		(.01)		(.00.)		(.02)		(.01)	
	0.05	0.04	0.05	0.02	0.05	0.03	0.05	0.03	0.03	0.03	-0.03	0.03
Log R&D expenditure	(.00.)	(.00)	(.00.)	(.00)	(.02)	(.00)	(.00.)	(.01)	(.00.)	(.00)	(.02)	(.00.)
Rural administrative	-0.12	-0.23	0.07	-0.08	-0.07	-0.06	-0.04	-0.07	-0.03	-0.13	0.04	-0.10
subordination	(.02)	(.01)	(.02)	(.01)	(.02)	(.01)	(.02)	(.01)	(.02)	(.01)	(.01)	(.01)
R-squared	0.76	0.90	0.42	0.80	0.41	0.78	0.84	0.56	0.62	0.83	0.57	0.84
Correlation between equations	0.22		0.13		0.09		0.12		0.13		0.13	
N	14,803		16,046		86,587		22,192		23,062		32,253	

Notes: All equations include a constant term and dummy variables for geographical location (22 provinces, 5 administrative regions and 4 municipalities, Beijing = reference), industry (38 categories, textile = reference), size (five categories, < 51 employees = reference), age (four categories, established earlier than 1994 = reference), unionization (1 if the firm has a union), the share of state capital in total capital, and the share of foreign and HKMT capital in total capital. Bold letters indicate that the null-hypothesis of equal wages and productivities for workers of gender i and skill-group j, $\lambda_i - \phi_i = 0$ or $\lambda_{ij} - \phi_{ij} = 0$, is rejected at the five percent level. Standard errors are reported in parenthesis.

5.3 Does non-wage compensation widen the gender-gap in earnings?

We now examine how non-wage compensation is distributed among male and female employees in the Chinese industrial sector. The sum of the firm's expenditures on social insurances is added to the wage measure in equation (4) before joint estimation with the production function (3). Results are placed in Table 3, columns (3) and (4). In particular, we are interested in comparing the parameters of the altered wage equation in column (4) to the benchmark salary-only estimates in column (1). This exercise shows that the distributions are highly similar. The earnings difference among unskilled workers remains unaffected, showing that non-wage benefits are not disproportionally allocated between men and women with less than 12 years of education.

Non-wage compensation is distributed to the advantage of skilled workers. It raises the average earnings received by skilled women by 17% (1.82-1.65), and those of skilled men by 8% (2.01-1.93). This result could reflect the incentives of firms to retain skilled workers by providing them with social insurances which are not easily transferred from one workplace to another. It could also be the consequence of a higher demand for tax exempt insurances among workers subjected to higher than average income tax levels. The result that skilled women are more likely than skilled men to be covered by social insurances means that the gender-gap in total earnings between skilled workers is narrower than the gap in wages. The male advantage in compensation is reduced from 29% to 19%.

Summing up the results for the role of social insurance payments in gender-distribution of earnings, we can conclude that such payments do not widen wage-disparities between unskilled men and women. Among workers with more than 12 years of education, payments are skewed slightly in favor of women, causing the gender-wage gap to become somewhat narrower than when we consider wages alone. The result that social insurance contributions are largely unimportant to the gender-distribution of work compensation is

robust to removing all firms that do not contribute to all the insurances considered²² (Table B1). Findings for the six ownership sectors are also robust to the inclusion of non-wage compensation in the earnings measure, as evidenced by comparing Table 4 to Table B2.

5.4 Robustness checks

A number of robustness checks carried out to assess the sensitivity of the main results presented above. We first replace the Cobb-Douglas production function with a translog version. Results placed in Table B3, Appendix B, show only slight alterations in the estimated earnings and productivity differentials in the full sample. Next, we consider if industrial segregation by gender is an important factor in gender-wage inequality, as suggested by some previous studies (Maurer-Fazio et al., 1999; Gustafsson and Li, 2000; Hughes and Maurer-Fazio, 2002). If women are crowded into low-paying industries, the female-to-male wage differential λ_F should rise if we remove the industrial dummies from the vector of control variables²³. Estimation results (not reported) however show no change in the parameter, meaning that industrial segregation does not substantially contribute to the gender-wage gap.

The use of cross-sectional data in this paper does not allow us to distinguish whether the lower productivity of females stem from their weaker performance within firms, or from a gender-productivity variation across firms. The latter could be the consequence of women being compelled to self-select into joining less productive enterprises. If women take jobs in firms that are more labor intensive, and labor has a decreasing return in the firm's production function, the share of women would be negatively correlated to firm productivity. We test this hypothesis by dividing the full sample according to the median capital-to-labor ratio. The estimation results do not provide support for the idea that women are sorted into more labor

²² this leaves 30,995 firms where total non-wage compensation make up on average 39% of the firms' total wage costs.

²³ The descriptive statistics in Table A3 indeed indicate substantial gender-segregation between industries. The share of female employees ranges from 12% in the 1,204 firms engaged in mining and processing of ferrous metal ores, to 70% in the 9,416 firms that manufacture apparel, footwear and caps.

intensive firms with lower labor productivity. The productivity differential between men and women (row one, Table B4) remains largely unchanged despite the sample division.

It has been argued that technological change that improves on firm productivity and conserves on labor leads to the dismissal of (generally male) production labor (Berman et al., 1994; Dunne et al., 1997). Under such circumstances, the productivity of females would be overestimated in our analysis, due to the positive correlation between the share of women and the productivity enhancing technological change. Table B5 examines this hypothesis by dividing the data into two sub-samples along the median share of women in the firm's workforce. Controlling for the share of females by splitting the sample in this way should yield larger gender differences in productivity if the hypothesis is true. Comparing the estimation results within the two sub-samples to the baseline in Table 3 shows that the skill-weighted average gender-difference in productivity remains constant.

If female employees work fewer hours than males, measuring labor input by the average number of female and male workers per year can lead to a negative bias in the estimated productivity of women. Using individual-level data, we can calculate labor input weights and attempt to correct this potential bias. The widely used China Household Income Project (CHIP) data for year 2002 includes information on the yearly, monthly and weekly work hours for employed men and women with different educational levels working for firms under different ownership. Average work-years for skilled and unskilled men and women are calculated by multiplying the usual number of hours worked in a week with the number of weeks worked in a month and the number of months worked in a year. Weights are calculated by normalizing the work-year measure for each group with respect to unskilled male workers.

The calculated labor input weights, presented in Table B6, are consistent with the expected gender and skill patterns. Employees with higher education work fewer hours per week than those with lower education. Female unskilled employees supply less labor than

their male unskilled colleagues in SOEs, but more in foreign-invested firms. Unambiguous ownership information in the CHIP data is not available for HKMT invested firms, or firms with the mixed ownership structures included in the category "Other" in this paper. Table B7 contains the estimation results with and without labor input weights after the exclusion of these two ownership sectors. The general conclusions of the paper concerning the productivity disadvantage of women remain unchanged. We however note that for skilled women and men, taking account of their on average shorter work-weeks compared to unskilled workers decreases the estimated under-payment of these groups by about 10 percentage points.

Labor unions are the bylaw designated monitors of firm compliance with antidiscriminatory laws and regulations and may therefore provide gender-specific wage protection (Cooke, 2001). Empirical research has provided evidence of more narrow genderearnings gaps in unionized organizations (i.e. Elvira and Saporta, 2001), and Ge (2007) showed that Chinese firms with established unions had higher levels of wages, non-wage benefits, and productivity. We examine the effect of unionization by estimating our wage- and production function in the sub-sample of 92,174 firms with unions, and that of 102,768 firms without unions, separately. The results in Table B8 show no effect of unionization on the average male-female wage or productivity gaps. For unskilled women, their productivity disadvantage is however slightly smaller (38% compared to 50%) in the unionized firms.

Dong and Zhang (2009) find that women are paid in excess of their productivity in the state-owned sector, but not in the private sector. They argue that the legacy from the planned economy is important to understanding this result: while the state assigned women to positions within heavy industry for which they were physically disadvantages, they were rewarded the same wages as men (Korzec, 1992). Results found in this paper do not provide support to Dong and Zhang's claim in the sense that the over-payment of women should be

restricted to the state-sector. In fact, the magnitude of the productivity disadvantage of women in SOEs is not statistically different from that in collective, private or HKMT-funded firms (see Table 4). However, the ownership-segregated results may be misinterpreted if privatized firms retain the legacy of mis-matched productivities and wages. Running the empirical analysis on a sub-sample of firms established more than 20 years ago (17,573 observations) however does not change the size of the gender-gap in productivity.

Regardless of the role of state planners in assigning women to specific jobs, female workers may still be "biologically" less productive than men in industries where a larger share of the jobs requires physical strength. Dividing the full sample into 14 aggregated industrial sectors²⁴ provides some support to this more general claim. Examining Table B9 we find that women's productivity disadvantages are larger than average in heavy industries such as the Power (75%), Petroleum (76%), and Chemical (57%) sectors.

China's special economic zones were set up to attract foreign investment and are the geographical locations where market reform has advanced the furthest. Table B10 contains the estimation results for a sample of 13,041 firms located in the six SEZs: Zhuhai and Shantou in Guangdong Province, Xiamen in Fujian Province, the entire province of Hainan, and Pudong New Area in the city of Shanghai. We first note that the estimated wage-gap between unskilled men and women drops from 11% in the full sample to 0%, while the wage gap among skilled workers remains constant. Examining the industry-specific wage-gaps in Table B9, we note that this result could stem from small gender-wage gaps among unskilled workers in light industrial sectors such as Tailoring (0%), Leather products (0%), and Textiles (6%). Recalling that these sectors are those with the largest shares of female workers, the conclusion is that market reform appears to have improved the prospects of gainful employment and the relative incomes of unskilled women²⁵.

²⁴ in accordance with the ex-factory price index categories used in China's official statistical publications.

²⁵ A note of caution when interpreting the positive effects on women's lives from the new employment opportunities is

From Table B10 also see that the productivity gap between unskilled men and women is not significantly different from zero in the Special Economic Zones. As a consequence, the hypothesis of equal wages and productivities for unskilled women cannot be rejected. One interpretation of this result is that firms in the Zones operate on more competitive product and labor markets, which pushes firms to organize labor more efficiently. Another interpretation is that expanded work opportunities in the light industrial sectors that dominate the SEZs have improved women's relative productivity.

6. Conclusions

Properly identifying the existence and extent of discrimination in work compensation is of great economic importance. Discriminated worker-groups are discouraged from participating in employment and from investing in skill development and education. As a consequence, human capital is underdeveloped and under-utilized, and long-term growth is negatively affected. This paper has used firm-level data for Chinese industrial enterprises to investigate if women face discriminatory behavior in the hands of employers.

The estimated gender-wage gap of 11% is narrower than those found for other large economies by researchers using firm-level data and similar methodologies. This result suggests that the Maoist ideology of gender equality has carried over in the market transition. Contradicting this hypothesis is the result that the narrowest gaps exist in the most liberalized sectors. Among firms that are located Special Economic Zones, and in certain light industrial sectors, women and men are estimated to earn on average equal wages. In light of these results, market reforms do not appear to have triggered greater wage inequality between men and women. Rather, the results suggest that a shift from heavy to light industry, combined

with labor market reforms, may have improved women's access to gainful employment. In line with this observation is the result that equal wages are observed in the industrial sectors with the highest shares of women, namely in Tailoring firms, and in firms producing Textile or Leather products. If women would be crowded into these sectors, then, in accordance with Bergmann's (1974) crowding-in model, we would expect relative wages to fall. In contrast, their expansion in the transition period appears to have improved women's relative incomes.

I find no empirical evidence of earnings discrimination against female industrial workers. On the contrary, the results suggest that women's wages exceed their contributions to firm productivity. However, a lower average productivity among female industrial workers could imply that female labor is less efficiently allocated than male labor. Comparing the size of the productivity disadvantage of women across the numerous sub-samples used in this paper shows that foreign-invested firms, and firms with HKMT-investments, have been more successful in harnessing women's productivity. Smaller productivity disadvantages of women are also found for firm located in China's Special Economic Zones. Female labor hence appears to have been more efficiently allocated in sectors that operate on product and labor markets where the competitive pressure is higher.

The results show a narrower gender-wage gap among workers with less than 12 years of education than among workers with higher educational attainments. A main contributor to this observation is that skilled men appear to have benefitted from the internationalization of industrial ownership to a greater extent than women.

Implementation of China's reformed social insurance system has been slow, and researchers have suggested the existence of a gender-bias in the distribution of the non-salary provisions. The findings in this paper do not support this conjecture. Estimation of a total earnings equation where each firm's expenditures on non-wage compensation are added to its total wage bill does not alter the estimated degree of overall gender polarization in earnings.

The under-representation of women in the generous State-owned firms hence appears to be balanced out by their over-representation in foreign-invested enterprises that are highly compliant with social insurance regulations.

References

Altonji, J. and Blank, R. (1999) 'Race and Gender in the Labor Market' in Ashenfelter, O and Card, D. Eds. *Handbook of Labor Economics*, vol. 3c. Elsevier Science, North Holland, 3144-3213.

Appleton, S., Knight, J, Song, L. and Xia, Q. (2002) 'Labor retrenchment in China: determinants and consequences', *China Economic Review* 13, 252-275.

Asano, H. and Kawaguchi, D. (2007) 'Male-Female Wage and Productivity Differentials: A Structural Approach Using Japanese Firm-level Panel Data, REITI Discussion Paper Series 07-E-020.

Banister, J. (2005) 'Manufacturing earnings and compensation in China', *Monthly Labor Review*, August 2005.

Bartolucci, C. (2008) 'Gender Wage Gaps Reconsidered: A Structural Approach Using Matched Employer-Employee Data', CEMFI Job Market Paper.

Bergmann, B. (1971) 'The Effect on White Incomes of Discrimination in Employment', Journal of Political Economy 79(2), 294-313.

Berman, E., Bound, J., and Griliches, Z. (1994) 'Changes in the Demand for Skilled Labor within U.S. Manufacturing: Evidence from the Annual Survey of Manufacturing', *Quarterly Journal of Economics* 109, 367-398.

Bian Y., Logan, J.R. and Shu, X. (2000) 'Economic Reform and the Gender Wage Gap in China' in Entwisle, B. and Henderson, G.E. Eds. *Re-Drawing Boundaries: Work, Household, and Gender in China*. Berkeley: University of California Press,111-33.

Bishop, J.A., Luo, F.J. and Wang, F. (2005) 'Economic Transition, Gender Bias, and the Distribution of Earning in Urban China', *Economics of Transition* 13(2), 239-260.

Berik, G., Dong, X. and Summerfield, G. (2007), 'China's transition and feminist economics', *Feminist Economics* 13 (3-4), 1-33.

Burda, J. (2007) 'Chinese women after the accession to the World Trade Organization: a legal perspective on women's labor rights', *Feminist Economics* 13 (3-4), 259-285.

Cao, Y. and Hu, C-Y. (2007) 'Gender and Job Mobility in Postsocialist China: A Longitudinal Study of Job Changes in Six Coastal Cities', Social Forces 85(4), 1535-1560.

Chinese Statistical Yearbook (2006) National Bureau of Statistics, Beijing: China Statistics Press.

Cooke, F. L. (2001) 'Equal opportunities? The role of legislation and public policies in women's employment in China', *Women in Management Review* 16(7), 334-348.

Croll, E. (1995) Changing Identities of Chinese Women: Rhetoric, Experience, and Self-perception in Twentieth Century China. London: Hong Kong UP ad Zed Press.

Démurger, S., Fournier, M. and Chen, Y. (2007) 'The Evolution of Gender Earnings Gaps and Discrimination in Urban China, 1988 and 1995', *Developing Economies* 45(1), 97-121.

Dong, X-Y. and Zhang, L. (2009) 'Economic transition and gender differentials in wages and productivity: Evidence from Chinese manufacturing enterprises', *Journal of Development Economics* 88, 144-156.

Dunne, T., Haltiwanger, J. and Troske, K. (1997) 'Technology and Jobs: Secular Change and Cyclical Dynamics', Carnegie-Rochester Conference Series on Public Policy 46.

Elvira, M. M. and Saporta, I. (2001) 'How does Collective Bargaining Affect the Gender Pay Gap?', *Work and Occupations* 28, 469 - 490.

Fleisher B. M. and Wang, X. (2004) 'Skill differentials, return to schooling, and market segmentation in a transition economy: the case of Mainland China', *Journal of Development Economics* 73, 315-328.

Ge, Y. (2007) 'What Do Unions Do in China?'. http://ssrn.com/abstract=1031084 (accessed January 27, 2009)

Giles, J., Park, A. and Cai, F. (2006) Re-employment of dislocates workers in urban China: the roles of information and incentives', *Journal of Comparative Economics* 34, 582-602.

Griliches, Z. and Ringstad, V. (1971) *Economies of Scale in the Form of the Production Function*, Amsterdam: North-Holland Publishing Company.

Gustafsson, B. and Li, S. (2000) 'Economic transformation and the gender earnings gap in urban China', *Journal of Population Economics* 13, 305-329.

Hellerstein J. K. and Neumark, D. (1999) 'Wages, and Productivity: An Empirical Analysis of Israeli Firm-Level Data', *International Economic Review* 40(1), 95-123.

Hellerstein, J. K, Neumark, D. and Troske, K.R. (1999) 'Wages, Productivity and Worker Characteristics: Evidence from Plant-Level Production Functions and Wage Equations', *Journal of Labor Economics* 17(3), 409-446.

Holz, C. and Lin, Y-M. (2001) 'Pitfalls of China's Industrial Statistics: Inconsistencies and Specification Problems', *The China Review* 1(1), 29-71.

Hughes, J. and Maurer-Fazio M. (2002) 'Effects of Marriage, Education and Occupation on the Female/Male Wage Gap in China', *Pacific Economic Review* 7(1), 137-156.

Jackson, R., Nakashima, K. and Neil H. (2009) 'China's long march to retirement reform', CSIS.

Jefferson, G.H, Rawski, T.G. and Zhang, Y. (2008) 'Productivity growth and the convergence across China's industrial economy', *Journal of Chinese Economy and Business Studies* 6(2), 121-140.

Jefferson, G.H. and Su, J. (2006) 'Privatization and restructuring in China: Evidence from shareholding ownership, 1995-2001', *Journal of Comparative Economics* 34(1), 146-166.

Korabik, K. (1994) 'Managerial women in the People's Republic of China: the long march continues' in Alder, N and Isralie, D. (eds) *Competitive frontiers: Women managers in the global economy*, Cambridge: Blackwell, 114-26.

Korzec, M. (1992) Labor and the Failure to Reform in China. St. Martin's Press: New York.

Liu, P-W, Meng, X and Zhang, J.S. (2000) 'Sectoral Gender Wage Differentials and Discrimination in the Transitional Chinese Economy', *Journal of Population Economics* 13, 311-352.

Liu, P-W, Zhang, J. and Ching Y.K. (2004) 'What Has Happened to the Gender Earnings Differential in Urban China during 1988–1999?' Hong Kong: Chinese University of Hong Kong. http://econrsss.anu.edu.au/pdf/china-abstract-pdf/Zhang-Kung.pdf (accessed December 23, 2009).

Liu, P-W., Meng, X. and Zhang, J.S. (2000) 'Sectoral Gender Wage Differentials and Discrimination in the Transitional Chinese Economy', Journal of Population Economics 13, 311-352.

Mauer-Fazio, M., Rawski, T., and Zhang, W. (1999) 'Inequality in the Rewards for Holding up Half the Sky: Gender Wage Gaps in China's Urban Labour Market, 1988-1994', *China Journal* 41, 55-88.

Maurer-Fazio, M. and Hughes, J. (2002) 'The effects of market liberalization on the relative earnings of Chinese women', *Journal of Comparative Economics* 30, 109-731.

Maurer-Fazio, M., Hughes, J. and Zhang, D. (2007) 'Gender, Ethnicity, and Labor Force Participation in Post-reform Urban China.' *Feminist Economics* 13(3–4), 189–212.

Millimet, D.L. and Wang, L. (2006), 'A Distributional Analysis of the Gender Earnings Gap in Urban China', *Contributions to Economic Analysis and Policy* 5(1), Article 5.

NBS (National Bureau of Statistics of China) (2006). China Statistical Abstract, Beijing: China Statistics Press.

Ngo, H.Y. (2002) 'Trends in Occupational Sex Segregation in Urban China', *Gender, Technology and Development* 6, 175-196.

Ng, Y.C. (2004), 'Economic Development, Human Capital, and Gender Earnings Differentials in China', *Economics of Education Review* 23(6), 587–603.

Peng, K. Z., Ngo, H-Y. and Wong, C-S. (2009) 'Gender differences in the work commitment of Chinese workers: An investigation of two alternative explanations', Journal of World Business 44(3), 323-335.

Pun, N. (2007) 'Gendering the dormitory labour system: production and reproduction and migrant labor in south China', *Feminist Economics* 13 (3-4), 239-58.

Razavi, S. (2007) 'Does Paid Work Enhance Women's Access to Welfare, Evidence from Selected Industrializing Countries', *Social Politics* 14(1), 58-92.

Rozelle, S., Dong, X., Zhang, L. and Mason, A. (2002) 'Gender Wage Gaps in Post-Reform Rural China', *Pacific Economic Review* 7(1), 157–179.

Sabin, L. (1994) 'New Bosses in the Workers' State: The Growth of Non-State Sector Employment in China, *The China Quarterly* 140, 994-970.

Stockman, N. (1994) 'Gender inequality and social structure in China', *Sociology* 28(3), 759-777.

Shen, J. (2007) Labor Disputes and their Management in China. Oxford: Chandos Publishing.

Shen, J. and Deng, X. (2008) 'Gender-wage inequality in the transitional Chinese economy: a critical review of post-reform research', *Journal of Organizational Transformation and Social Change* 5(2), 109-126.

Shu, X.L. and Bian, Y.J. (2003) 'Market Transition and Gender gap in Earnings in Urban China', Social Forces 81(4), 1107-1145.

Song, Y. and Dong, X-Y. (2009) 'Gender and Occupational Mobility in Urban China during the Economic Transition', Mimeo online: http://www.cwe.org.cn/info/UserFiles/2009-11/24/2009112414915412.pdf (accessed January 27, 2009).

Tan, S. (2000) 'The Relationship between Foreign enterprises, Local Governments, and Women Migrant Workers in the Pearl River Delta', in West, L. and Yaohui Z. (eds.) *Rural Labor Flows in China*, Berkeley: Institute of East Asian Studies, University of California, 292-309.

Vodopivec, M. and Tong M.H. (2008) 'China: Improving Unemployment Insurance', World Bank SP Discussion Paper no. 0820.

Wang, J. (2006) 'Health Sector Reform in China: Gender, Equality and Social Justice." In *Gender and Social Policy in a Global Context: Uncovering the Gendered Structure of 'the Social*', eds. S. Razavi, and S. Hassim. Basingstoke, UK: Palgrave 2006.

Wang M. and Cai, F. (2008), 'Gender Earnings Differential in Urban China', *Review of Development Economics* 12(2), 442-454.

Wang, Y-P, Wang, Y and Bramley, G. (2005) 'Chinese Housing Reform in State-owned Enterprises and Its Impacts on Different Social Groups', *Urban Studies* 42(10), 1859–1878.

Xu, L., Wang, Yan, C, Charles, D., and Shenglan, T. (2007) 'Urban health insurance reform and coverage in China using data from national Health Services Surveys in 1998 and 2003', *BMC Health Services Research* 7(37), 1-14.

Xu, W., Tan, K-C., and Wang, G.X. (2006), 'Segmented Local Labour Markets in Post-Reform China: Gender Earnings Inequality in the Case Of Two Towns in Zhejiang Province', *Environment and Planning A* 38, 85–109.

Appendix A: Additional descriptive statistics

Table A1 Components of Chinese urban earnings statistics

The statistical concept of wage (gongzi) or earnings for on-post urban "staff and workers" includes the					
following components, whether the employees receive the earnings or benefits in money or in kind and whether					
the earnings or benefits are or are not taxable items:					
Monthly or annual salary income (including base	Transport subsidy (car or shuttle bus provided, cash				
earnings and additions based on position, seniority,	for bus or taxi, and so on)				
wage scale, and so on)	Housing subsidy (dormitory provided, or directly				
Earnings during on-the-job training, probationary	Subsidized rent or purchase of housing)				
period	Individual income tax deducted from earnings and				
Employee income paid on an irregular basis	paid directly by enterprise to government				
Hourly payment for work performed	Social insurance funds (pension, medical,				
Piecework payment for work performed	unemployment insurance funds, and housing				
Bonus payments	purchase fund) deducted from the employee's				
Incentive, performance-based payments	wage and paid by the work unit to government				
Overtime pay	on behalf of the employee				
Hardship, danger pay	Money for rent, and utilities (electricity, water)				
All kinds of subsidies in cash or in kind	Money given for fixed line or mobile phone				
Festival, holiday subsidy	Clothing subsidy				
Travel money, food allowance while traveling	Subsidy compensating workers for lack of vacation				
Transport subsidy (car or shuttle bus provided, cash	time				
for bus or taxi, and so on)	Earnings during approved leaves of absence, pay for				
Personal services such as baths, haircuts	time not worked (regular vacation, compassionate				
Books, newspapers, magazines provided for	leave, to visit relatives, family-planning operation,				
employees	national or societal duty, study leave, leave due to				
Meals provided, food allowance	sickness or injury)				
Earnings during on-the-job training, probationary	Anything that has the nature or spirit of labor				
period	earnings, even if it is not spelled out in the				
Earnings during on-the-job training, probationary	regulations				
period					

Source: *Laodong gongzi; tongji taizhang* [*Labor wages; statistical accounts*] (Beijing, Beijing Municipality Statistical Bureau, 2004), pp. 2–1 to 2–5, cited in Banister (2005)

Table A2
Aggregation of registration-based classification of firms into ownership categories

riggiogation of registration basea e	Categories under which data on all industrial
Registration-based classification	enterprises are reported in the Statistical
(since 1998)	Yearbook
Domestic enterprises	
SOEs	State
COEs	Collective
Employee shareholding company	Collective
Joint operation enterprises	
State-owned	State (1)
Collective owned	Other
State- and collective owned	Other (2)
Other joint operation enterprises	Other (2)
Limited liability companies	
Solely state-owned	State
Others	State or Other (3)
Stock companies	State or Other (3)
Private enterprises	
Private sole proprietorships	Private
Private partnerships	Private
Private limited liability company	Private
Private stock companies	Private
Other enterprises	Other
HKMT-invested enterprises	
Joint equity ventures (JVEs)	HKMT or State (2)
Contractual joint ventures (CJVs)	HKMT or State (2)
Wholly HKMT-owned	HKMT
HKMT stock companies	HKMT or State (2)
Foreign-invested enterprises	
Chinese-foreign JEVs	Foreign or State (2)
Chinese-foreign CJVs	Foreign or State (2)
Wholly foreign-owned	Foreign
Foreign-invested stock company	Foreign or State (2)

Notes: The table details the ownership aggregation of registration-based firm classification system into six broader categories when reporting data on all industrial firms in the China Statistical Yearbook. Departures from the CSY aggregation method are reported in footnotes (1)-(3) below. Information about the CSY aggregation methodology is contained in Holz and Lin (2001).

- (1) Unlike in the Statistical Yearbook, this category is not double-counted as "Other".
- (2) When ownership-disaggregated statistics are reported per ownership in the Statistical Yearbook, a proportion of the statistic is double-counted in the "State" category. This proportion corresponds to the aggregate share of the sum of state capital to the sum of total paid-in capital minus individual capital in the registration-based classification category. I instead count an individual firm as State-owned if the state's share in total capital is greater than 50%.
- (3) This category is counted as "State" in the Statistical Yearbook for firms that are under absolute state control (guoyou juedui konggu) or relative state control (guoyou xiahgdui konggu). The first implies that the state account for more than 50% of total capital. The second that the state holds less than 50% of total capital but that i) its share is relatively large compared to the shares of other ownership categories, or ii) even though one or more other ownership categories have a larger capital share, the state in effect holds the control right s by agreement (Xiyi kongzhi). In this paper, only absolute state-controlled firms may be identified, and these are moved from the "Other" to the "State" category.

Table A3
Number of firms and share of female employees by 2-digit industrial sector

		Share of
	Number of	Female
2-digit Industry	Firms	Employee
Manufacturing Industry		
Mfg. of Apparel, Footwear and Caps	9,416	0.70
Mfg. of Articles for Culture, Education and Sports	2,754	0.58
Mfg. of Artwork and Other Manufacturing	272	0.27
Mfg. of Beverages	2,265	0.42
Mfg. of Chemical Fiber	1,040	0.46
Mfg. of Comm. Equip., Computers, and Electronic Equipment	2,884	0.44
Mfg. of Electrical Machinery and Equipment	7,136	0.53
Mfg. of Foods	3,809	0.49
Mfg. of Furniture	2,236	0.33
Mfg. of General Purpose Machinery	14,772	0.28
Mfg. of Articles for culture, education and Sports	3,920	0.57
Mfg. of Leather, Fur, Feather and Related Products	4,846	0.59
Mfg. of Medicines	3,316	0.46
Mfg. of Metal Products	10,552	0.33
Mfg. of Non-metallic Mineral Products	14,064	0.29
Mfg. of Paper and Paper Products	5,443	0.39
Mfg. of Plastics	9,198	0.45
Mfg. of Rubber	2,254	0.44
Mfg. of Special Purpose Machinery	7,245	0.27
Mfg. of Tobacco	140	0.40
Mfg. of Transport Equipment	8,569	0.31
Mfg. of Weapons and Ammunition	11,898	0.43
Mfg. of raw Chemical Mat'ls and Chem. Products	13,718	0.30
Textile Industry	17,484	0.64
Jining and Utilities	,	
Extraction of Petroleum and Natural Gas	142	0.19
Mining and Processing of Ferrous Metal Ores	1,204	0.12
Mining and Processing of Non-Ferrous Metal Ores	950	0.17
Mining and Processing of Non-metal Ores	1,500	0.17
Mining and Washing of Coal	3,406	0.10
Mining and Processing of Other Ores	10	0.15
Printing, Reproduction and Recording Media	3,772	0.45
Food Processing	9,384	0.38
Processing of Petroleum and Nuclear Fuel	1,357	0.24
Processing of Timber, Mfg. of Wood, etc. Products	3,618	0.43
Production and Distrib. of Water	2,183	0.40
Production and Distrib. of Electric Power and Heat Power	4,348	0.28
Production and Distrib. of Gas	369	0.32
Smelting and Pressing of Ferrous Metals	4,722	0.19
Smelting and Pressing of Non-ferrous Metals	3,644	0.25

Table A4
Number of firms and share of female employees by geographical region

14umoci of mins and	u snare or re	mare employe
		Share of
Province, Municipality	Number of	Female
or Autonomous Region	Firms	Employees
Municipalities		
Beijing	4,619	0.35
Chongqing	2,089	0.35
Shanghai	12,870	0.42
Tianjin	3,931	0.37
Autonomous Regions		
Guanxi	2,558	0.37
Inner Mongolia	1,824	0.31
Ningxia	432	0.32
Xinjiang	989	0.33
Tibet	136	0.26
Provinces		
Anhui	3,812	0.39
Fujian	10,091	0.46
Gansu	1,121	0.30
Guangdong	28,419	0.44
Guizhou	1,533	0.29
Hainan	431	0.37
Hebei	7,11	0.33
Heilongjiang	1,901	0.34
Henan	114	0.36
Hubei	4,461	0.38
Hunan	6,001	0.33
Jiangsu	28,842	0.45
Jiangxi	2,872	0.41
Jilin	1,92	0.33
Liaoning	7,681	0.34
Qinghai	256	0.29
Shaanxi	2,042	0.32
Shandong	18,957	0.40
Shanxi	3,043	0.20
Sishuan	5,577	0.31
Yunnan	1,644	0.31
Zhejiang	32,564	0.43

Appendix B: Robustness checks

Table B1

Joint production function and earnings equation estimates: Cobb-Douglas production function, firms with non-zero expenditures on all three non-wage compensation items

Tunction, in his with hon zero	on pontarion	911 0111 0111 00	more wage es	inpensation iten
			Log	Log
Wages, total compensation, and	Log (Value	Log	(Value	(Total
productivities relative to unskilled	Added)	(Wage)	Added)	Compensation)
male workers for	(1)	(2)	(3)	(4)
	0.62	0.89	0.62	0.90
female workers	(.04)	(.02)	(.04)	(.02)
	0.71	0.88	0.71	0.88
female unskilled workers	(.05)	(.05)	(.05)	(.02)
	2.30	1.80	2.28	1.86
female skilled workers	(.24)	(.07)	(.23)	(.07)
	4.51	2.14	4.45	2.07
male skilled workers	(.27)	(.05)	(.27)	(.05)
Wage-productivity gap for				
female workers	0.26		0.28	
female unskilled workers	0.15		0.17	
female skilled workers	-0.50		-0.42	
male skilled workers	-2.38		-2.37	
	0.23		0.27	
Log capital	(.00)		(.00)	
	0.70		0.71	
Log labor	(.01)		(.05)	
	0.04	0.02	0.04	0.02
Log R&D expenditure	(.00)	(.00.)	(.00)	(.00)
_	0.01	-0.14	0.01	-0.16
Rural administrative subordination	(.01)	(.01)	(.02)	(.01)
R-squared	0.66	0.87	0.86	0.67
Correlation between equations	0.15		0.15	
N	30,995		30,995	

Table B2

Joint production function and total compensation equation estimates: Cobb-Douglas production function, ownership-based sub-samples

	SOE COE		PRI HKMT			FIE Oth						
	30	E	CO	E	PR	.1	пкг	V1 1	FI	C I	Oth	ICI
Total compensation and productivities relative to unskilled male workers for	Log (Value Added)	Log (Total Comp.)	Log (Value Added))	Log (Total Comp.)	Log (Value Added)	Log (Total Comp.)	Log (Value Added)	Log (Total Comp.)	Log (Value Added)	Log (Total Comp.)	Log (Value Added)	Log (Total Comp.)
	0.58	0.92	0.54	1.02	0.53	0.90	0.65	0.86	0.71	0.83	0.57	0.93
female workers	(.06)	(.03)	(.06)	(.02)	(.02)	(.01)	(.03)	(.01)	(.03)	(.02)	(.04)	(.02)
	0.70	0.82	0.42	0.93	0.49	0.90	0.69	0.92	0.71	0.96	0.52	0.91
female unskilled workers	(.08)	(.04)	(.05)	(.02)	(.02)	(.01)	(.04)	(.02)	(.04)	(.02)	(.04)	(.02)
	2.30	2.00	3.10	2.40	1.98	1.37	2.61	1.77	3.81	2.17	2.24	1.80
female skilled workers	(.38)	(.11)	(.61)	(.14)	(.17)	(.04)	(.26)	(.08)	(.31)	(.09)	(.25)	(.07)
	4.98	1.82	3.10	1.37	2.91	1.50	4.65	2.59	5.85	3.52	3.62	1.79
male skilled workers	(.44)	(.08)	(.46)	(.08)	(.15)	(.03)	(.34)	(.08)	(.42)	(.10)	(.24)	(.04)
Wage-productivity gap for												
female workers	0.23		0.47		0.36		0.22		0.12		0.37	
female unskilled workers	0.12		0.51		0.41		0.23		0.24		0.39	
female skilled workers	-0.29		-0.49		-0.61		-0.85		-1.64		-0.44	
male skilled workers	-3.15		-1.72		-1.41		-2.06		-2.33		-1.83	
	0.20		0.20		0.21		0.19		0.21		0.22	
Log capital	(.01)		(.01)		(.00)		(.00)		(.00)		(.00)	
	0.77		0.50		0.53		0.71		0.71		0.65	
Log labor	(.02)		(.02)		(.01)		(.02)		(.04)		(.01)	
	0.05	0.05	0.05	0.03	0.05	0.04	0.05	0.03	0.03	0.03	0.04	0.04
Log R&D expenditure	(.00)	(.02)	(.01)	(.00)	(.00)	(.00)	(.00)	(.00)	(.00)	(.00)	(.00)	(.00)
Rural administrative	-0.13	-0.29	0.07	-0.13	-0.07	-0.07	-0.04	-0.08	-0,03	-0.14	-0.04	-0.11
subordination	(.02)	(.01)	(.01)	(.01)	(.02)	(.01)	(.02)	(.31)	(.02)	(.01)	(.01)	(.01)
R-squared	0.76	0.89	0.42	0.78	0.41	0.76	0.56	0.83	0.90	0.53	0.57	0.83
Correlation between equations			0.13		0.10		0.12		0.13		0.13	
N	14,805		16,046		86,591		22,192		20,063		32,253	

Table B3
Joint production function and earnings equation estimates:

Translog production function, all firms

Wages, total compensation, and productivities relative to unskilled male workers for	Log (Value Added)	Log (Wages)	Log (Value Added)	Log (Total Compensation)
	0.54	0.90	0.54	0.89
female workers	(.01)	(.01)	(.01)	(.01)
	2.57	1.66	2.56	1.83
female unskilled workers	(.12)	(.03)	(.12)	(.03)
	3.98	1.93	3.95	2.02
female skilled workers	(.12)	(.02)	(.12)	(.02)
male skilled workers	` '		` /	, ,
Wage-productivity gap for	0.30		0.31	
female workers	0.36		0.35	
female unskilled workers	-0.91		-0.73	
female skilled workers	-2.04		-1.93	
	0.39		0.40	
female workers	(.02)		(.02)	
	-0.17		-0.17	
female unskilled workers	(.01)		(.01)	
	-0.02		-0.02	
female skilled workers	(.00.)		(.00)	
	0.04		0.04	
male skilled workers	(.00.)		(.00)	
	0.03		0.03	
Wage-productivity gap for	(.00)		(.00)	
	0.04	0.03	0.04	0.04
female workers	(.00)	(.00.)	(.00)	(.00)
	-0.04	-0.12	-0.04	-0.15
female unskilled workers	(.01)	(.00)	(.01)	(.00)
female skilled workers	0.57	0.84	0.57	0.83
male skilled workers	0.12		0.13	
N	194,942		194,950	

Table B4
Joint production function and earnings equation estimates: Cobb-Douglas production function, firm sub-samples based on the median capital-to-labor ratio

production function, firm sub-sa	impies based on	me mear	an capitai-to-ia	oor rauo		
	High capital-to-l	abor ratio,	Low capital-to-lab	Low capital-to-labor ratio,		
	above or equal to	o median	below median (1.7	76)		
	(1.76)					
Wages and productivities relative to unskilled male workers for	Log (Value Added)	Log (Wages)	Log (Value Added)	Log (Wages)		
	0.59	0.92	0.64	0.93		
female workers	(.03)	(.01)	(.01)	(.01)		
	0.55	0.93	0.61	0.93		
female unskilled workers	(.03)	(.03)	(.02)	(.01)		
	2.31	1.61	2.46	1.55		
female skilled workers	(.17)	(.04)	(.14)	(.04)		
	3.86	1.88	3.47	1.73		
male skilled workers	(.17)	(.03)	(.13)	(.03)		
Wage-productivity gap for						
female workers	0.33		0.29			
female unskilled workers	0.38		0.32			
female skilled workers	-0.70		-0.90			
male skilled workers	-1.98		-1.74			
	0.28		0.16			
Log capital	(.00)		(.00)			
	0.53		0.73			
Log labor	(.01)		(.01)			
	0.04	0.03	0.0	0.03		
Log R&D expenditure	(.00)	(.00.)	(.00)	(.00)		
_	-0.08	-0.13	-0.02	-0.11		
Rural administrative subordination	(.01)	(.01)	(.01)	(.00)		
R-squared	0.52	0.79	0.60	0.86		
Correlation between equations	0.13		0.12			
N	97,122		97,820			

Table B5
Joint production function and earnings equation estimates: Cobb-Douglas production function, firm sub-samples based on the median share of female employees

High share of females, above median (0.37)	Tunction, firm sub-samples ba					
Wages and productivities relative to unskilled male workers for Log (Value Added) Log (Wages) Log (Ualue Added) Log (Wages) Log (Ualue Added) Log (Day)			ales, above			
Wages and productivities relative to unskilled male workers for (Value Added) (Wages) (Value Added) (Wages) female workers 0.59 0.86 0.61 1.06 female workers 0.67 0.94 0.42 0.89 female unskilled workers (.03) (.01) (.05) (.02) female skilled workers (.15) (.04) (.24) (.06) male skilled workers (.27) (.06) (.13) (.03) Wage-productivity gap for (.27) (.06) (.13) (.03) female workers 0.25 0.45 (.03) female unskilled workers 0.26 0.47 (.03) female unskilled workers 0.26 0.47 (.027 male skilled workers 0.26 0.47 (.027 male skilled workers 0.26 0.23 (.00) gap and the workers 0.03 0.04 0.03 gap and the workers 0.05 0.03 0.04 0.03		median (0.37)		or equal to median (0.37)		
Wages and productivities relative to unskilled male workers for (Value Added) (Wages) (Value Added) (Wages) female workers 0.59 0.86 0.61 1.06 female workers 0.67 0.94 0.42 0.89 female unskilled workers (.03) (.01) (.05) (.02) female skilled workers (.15) (.04) (.24) (.06) male skilled workers (.27) (.06) (.13) (.03) Wage-productivity gap for (.27) (.06) (.13) (.03) female workers 0.25 0.45 (.03) (.04) (.04) (.04) (.04) (.04) (.04) (.04) (.04) (.06) (.03) (.06) (.13) (.06) (.03) (.06) (.13) (.03) (.06) (.03) (.04) (.04) (.04) (.04) (.04) (.04) (.04) (.04) (.04) (.05) (.04) (.04) (.05) (.04) (.05) (.04) <t< td=""><td></td><td></td><td></td><td></td><td>r</td></t<>					r	
Wages and productivities relative to unskilled male workers for (Value Added) (Wages) (Value Added) (Wages) female workers 0.59 0.86 0.61 1.06 female workers 0.67 0.94 0.42 0.89 female unskilled workers (.03) (.01) (.05) (.02) female skilled workers (.15) (.04) (.24) (.06) male skilled workers (.27) (.06) (.13) (.03) Wage-productivity gap for (.27) (.06) (.13) (.03) female workers 0.25 0.45 (.03) female unskilled workers 0.26 0.47 (.03) female unskilled workers 0.26 0.47 (.027 male skilled workers 0.26 0.47 (.027 male skilled workers 0.26 0.23 (.00) gap and the workers 0.03 0.04 0.03 gap and the workers 0.05 0.03 0.04 0.03		Log	Log	Log	Log	
Second Color Seco	Wages and productivities relative to		0		_	
female workers (.02) (.01) (.05) (.02) female unskilled workers (.03) (.01) (.05) (.02) female skilled workers (.15) (.04) (.24) (.06) female skilled workers (.27) (.06) (.13) (.03) Wage-productivity gap for (.27) (.06) (.13) (.03) female workers 0.25 0.45 (.03) female unskilled workers 0.26 0.47 (.04) (.04) (.04) (.04) (.03) female workers (.27) (.06) (.13) (.03) (.03) (.03) (.03) (.03) (.03) (.03) (.03) (.04) (.03) (.04) (.04) (.05) (.04) (.04) (.05) (.04) (.04) (.04) (.04) (.04) (.04) (.05) (.04) (.04) (.05) (.04) (.05) (.06) (.06) (.06) (.06) (.06) (.06) (.06) (.0	unskilled male workers for	(Value / Idded)	(Wages)	(value ridded)	(Wages)	
0.67 0.94 0.42 0.89		0.59	0.86	0.61	1.06	
female unskilled workers (.03) (.01) (.05) (.02) female skilled workers (.15) (.04) (.24) (.06) male skilled workers (.27) (.06) (.13) (.03) Wage-productivity gap for (.27) (.06) (.13) (.03) female workers 0.25 0.45 (.03) female unskilled workers 0.26 0.47 (.04) female skilled workers 0.26 0.45 (.04) male skilled workers -1.11 -0.27 (.07) male skilled workers -2.66 -1.73 (.00) male skilled workers -2.66 -1.73 (.00) male skilled workers -0.20 0.23 (.00) male skilled workers -0.00 (.00) (.00) male skilled workers -0.03 0.04 0.03 getable skilled workers -0.03 0.04 0.00 getable skilled workers -0.03 0.04 0.03	female workers	(.02)	(.01)	(.05)	(.02)	
2.51		0.67	0.94	0.42	0.89	
female skilled workers (.15) (.04) (.24) (.06) male skilled workers 5.01 2.36 3.35 1.61 male skilled workers (.27) (.06) (.13) (.03) Wage-productivity gap for female workers 0.25 0.45 female workers 0.26 0.47 female skilled workers -1.11 -0.27 male skilled workers -2.66 -1.73 Log capital (.00) (.00) Log capital (.00) (.00) Log labor (.01) (.01) Log R&D expenditure (.00) (.00) Log R&D expenditure (.00) (.00) -0.05 -0.12 -0.07 -0.13 Rural administrative subordination (.01) (.00) (.02) (.00) R-squared 0.54 0.84 0.57 0.84 Correlation between equations 0.12 0.14 0.14	female unskilled workers	(.03)	(.01)	(.05)	(.02)	
S.01 2.36 3.35 1.61		2.51	1.41	2.68	2.42	
male skilled workers (.27) (.06) (.13) (.03) Wage-productivity gap for female workers 0.25 0.45 female unskilled workers 0.26 0.47 female skilled workers -1.11 -0.27 male skilled workers -2.66 -1.73 male skilled workers 0.20 0.23 Log capital (.00) (.00) Log labor (.01) (.01) Log labor (.01) (.01) Log R&D expenditure (.00) (.00) (.00) Log R&D expenditure (.00) (.00) (.00) Rural administrative subordination (.01) (.00) (.02) (.00) R-squared 0.54 0.84 0.57 0.84 Correlation between equations 0.12 0.14	female skilled workers	(.15)	(.04)	(.24)	(.06)	
Wage-productivity gap for 0.25 0.45 female workers 0.26 0.47 female unskilled workers -1.11 -0.27 male skilled workers -2.66 -1.73 male skilled workers -2.66 -1.73 male skilled workers 0.20 0.23 male skilled workers 0.02 0.02 male skilled workers 0.02 0.00 male skilled workers 0.02 0.00 male skilled workers 0.02 0.00 male skilled workers 0.00 0.00 male skilled workers 0.00 0.00 male skilled workers 0.02 0.00 male skilled workers <		5.01	2.36	3.35	1.61	
female workers 0.25 0.45 female unskilled workers 0.26 0.47 female skilled workers -1.11 -0.27 male skilled workers -2.66 -1.73 male skilled workers 0.20 0.23 Log capital (.00) (.00) Log labor (.01) (.01) Log labor 0.05 0.03 0.04 0.03 Log R&D expenditure (.00) (.00) (.00) (.00) Log R&D expenditure (.00) (.00) (.00) (.00) Rural administrative subordination (.01) (.00) (.02) (.00) R-squared 0.54 0.84 0.57 0.84 Correlation between equations 0.12 0.14 0.14	male skilled workers	(.27)	(.06)	(.13)	(.03)	
female unskilled workers 0.26 0.47 female skilled workers -1.11 -0.27 male skilled workers -2.66 -1.73 0.20 0.23 Log capital (.00) (.00) Log labor (.01) (.01) Log R&D expenditure (.00) (.00) (.00) Log R&D expenditure (.00) (.00) (.00) Rural administrative subordination (.01) (.00) (.02) (.00) R-squared 0.54 0.84 0.57 0.84 Correlation between equations 0.12 0.14	Wage-productivity gap for					
female skilled workers -1.11 -0.27 male skilled workers -2.66 -1.73 0.20 0.23 Log capital (.00) (.00) 0.63 0.62 Log labor (.01) (.01) 0.05 0.03 0.04 0.03 Log R&D expenditure (.00) (.00) (.00) (.00) Rural administrative subordination (.01) (.00) (.02) (.00) R-squared 0.54 0.84 0.57 0.84 Correlation between equations 0.12 0.14	female workers	0.25		0.45		
male skilled workers -2.66 -1.73 Log capital 0.20 0.23 Log capital (.00) (.00) Log labor 0.63 0.62 Log labor (.01) (.01) Log R&D expenditure (.00) (.00) (.00) Log R&D expenditure (.00) (.00) (.00) Rural administrative subordination (.01) (.00) (.02) (.00) R-squared 0.54 0.84 0.57 0.84 Correlation between equations 0.12 0.14 0.14	female unskilled workers	0.26		0.47		
Log capital 0.20 (.00) 0.23 (.00) Log labor 0.63 (.01) 0.62 (.01) Log R&D expenditure 0.05 (.00) (.00) 0.03 (.00) 0.04 (.00) Log R&D expenditure 0.00 (.00) (.00) (.00) 0.00 0.00 Rural administrative subordination 0.01 (.00) (.00) (.02) (.00) 0.00 R-squared 0.54 (0.84 (0.84 (0.57) (0.04)) 0.57 (0.84) Correlation between equations 0.12 (0.14) 0.14	female skilled workers	-1.11		-0.27		
Log capital (.00) (.00) Log labor (.01) (.01) Log R&D expenditure (.00) (.00) (.00) Log R&D expenditure (.00) (.00) (.00) Log R&D expenditure (.00) (.00) (.00) Rural administrative subordination (.01) (.00) (.02) R-squared 0.54 0.84 0.57 0.84 Correlation between equations 0.12 0.14 0.14	male skilled workers	-2.66		-1.73		
Log labor 0.63 (.01) 0.62 (.01) Log R&D expenditure 0.05 (.00) 0.03 (.00) 0.00) Log R&D expenditure (.00) (.00) (.00) (.00) (.00) (.00) Rural administrative subordination (.01) (.00) (.00) (.02) (.00) R-squared 0.54 (0.84) (0.84) (0.57) 0.84 Correlation between equations 0.12 0.14		0.20		0.23		
Log labor (.01) (.01) 0.05 0.03 0.04 0.03 Log R&D expenditure (.00) (.00) (.00) (.00) -0.05 -0.12 -0.07 -0.13 Rural administrative subordination (.01) (.00) (.02) (.00) R-squared 0.54 0.84 0.57 0.84 Correlation between equations 0.12 0.14 0.14	Log capital	(.00)		(.00)		
Log R&D expenditure 0.05 (.00) 0.03 (.00) 0.04 (.00) 0.03 (.00) Log R&D expenditure (.00) (.00) (.00) (.00) (.00) Rural administrative subordination (.01) (.00) (.02) (.00) R-squared 0.54 0.84 0.57 0.84 Correlation between equations 0.12 0.14		0.63		0.62		
Log R&D expenditure (.00) (.00) (.00) (.00) -0.05 -0.12 -0.07 -0.13 Rural administrative subordination (.01) (.00) (.02) (.00) R-squared 0.54 0.84 0.57 0.84 Correlation between equations 0.12 0.14	Log labor	(.01)		(.01)		
-0.05 -0.12 -0.07 -0.13 Rural administrative subordination (.01) (.00) (.02) (.00) R-squared 0.54 0.84 0.57 0.84 Correlation between equations 0.12 0.14		0.05	0.03	0.04	0.03	
Rural administrative subordination (.01) (.00) (.02) (.00) R-squared 0.54 0.84 0.57 0.84 Correlation between equations 0.12 0.14 0.14	Log R&D expenditure	(.00)	(.00)	(.00)	(.00)	
R-squared 0.54 0.84 0.57 0.84 Correlation between equations 0.12 0.14	_	-0.05	-0.12	-0.07	-0.13	
Correlation between equations 0.12 0.14	Rural administrative subordination	(.01)	(.00.)	(.02)	(.00.)	
	R-squared	0.54	0.84	0.57	0.84	
N 97.598 97.344	Correlation between equations	0.12		0.14		
	N	97,598		97,344		

Table B6
Labor input weights for gender-skill groups by firm ownership

	out weights for go	7110-01 511111	8-04-ps 25 22	o ;;	
		Labor			Labor
Ownership	Gender-Skill	Input	Ownership	Gender-Skill	Input
Sector	Group	Weight	Sector	Group	Weight
SOE	Unskilled males	1.000	PRI	Unskilled males	1.000
	Unskilled females	0.959		Unskilled females	0.955
	Skilled males	0.992		Skilled males	0.918
	Skilled females	0.945		Skilled females	0.738
COE	Unskilled males	1.000	FIE	Unskilled males	1.000
	Unskilled females	0.951		Unskilled females	1.103
	Skilled males	0.999		Skilled males	0.945
	Skilled females	0.861		Skilled females	0.964

Table B7
Joint production function and earnings equation estimates:
Cobb-Douglas production function, gender-skill weighted labor inputs

3 1	Tailerion, gender binir weighted tabor inputs					
	No weig	hts	Weights			
Wages and productivities relative to unskilled male workers for	Log (Value Added)	Log (Wages)	Log (Value Added)	Log (Wages)		
	0.58	0.91	0.58	0.91		
female workers	(.02)	(.01)	(.02)	(.01)		
female unskilled workers	0.55 (.02)	0.90 (.01)	0.55 (.02)	0.89 (.01)		
female skilled workers	2.15 (.13)	1.53 (.03)	2.15 (.13)	1.71 (.03)		
	3.51	1.70	3.48	1.77		
male skilled workers	(.12)	(.02)	(.12)	(.02)		
Wage-productivity gap for						
female workers	0.33		0.33			
female unskilled workers	0.35		0.34			
female skilled workers	-0.61		-0.44			
male skilled workers	-1.81		-1.70			
Log capital	0.21 (.00)		0.21 (.00)			
Log labor	0.60 (.01)		0.60 (.01)			
Log R&D expenditure	0.05 (.00)	0.03 (.00)	0.05 (.00)	0.04 (.00)		
	-0.08	-0.13	-0.08	-0.16		
Rural administrative subordination	(.01)	(.00.)	(.01)	(00.)		
R-squared	0.94	0.49	0.94	0.51		
Correlation between equations	0.12		0.13			

Table B8
Joint production function and earnings equation estimates
Cobb-Douglas production function, firms with and without labor union

Cobb-Douglas production function, in his with and without labor union							
	Unio	on	No uni	No union			
Wages and productivities relative to unskilled male workers for	Log (Value Added)	Log (Wages)	Log (Value Added)	Log (Wages)			
female workers	0.59 (.02)	0.87 (.01)	0.58 (.02)	0.90 (.01)			
female unskilled workers	0.62 (.02)	0.90 (.01)	0.50 (.02)	0.90 (.01)			
female skilled workers	2.49 (.17)	1.76 (.05)	5.52 (.15)	1.59 (.03)			
male skilled workers	4.63 (.17)	2.24 (.04)	3.48 (.14)	1.75 (.03)			
Wage-productivity gap forfemale workers	0.28		0.32				
female workers	0.28		0.32				
female skilled workers	-0.73		-0.93				
male skilled workers	-2.39		-1.72				
Log capital	0.24 (.00) 0.69		0.20 (.00) 0.55				
Log labor	(.01)		(.01)				
Log R&D expenditure	0.05 (.01)	0.03 (.00)	0.04 (.00)	0.03 (.00)			
Rural administrative	-0.06	-0.14	-0.04	-0.08			
subordination	(.01)	(.00.)	(.01)	(.01)			
R-squared	0.62	0.86	0.46	0.80			
Correlation between equations	0.14		0.11				
N	92,174		102,768				

Table B9

Joint production function and earnings equation estimates: Industrial categories

Joint production		on una cu	8			1114454114	cutegori	
	Metal	lurgical	Po	ower	C	coal	Petro	oleum
Wages and								
productivities								
relative to	Log		Log		Log		Log	
unskilled male	(Value	Log	(Value	Log	(Value	Log	(Value	Log
workers for	Added)	(Wages)	Added)	(Wages)	Added)	(Wages)	Added)	(Wages)
	0.45	0.85	0.25	0.73	2.43	0.87	0.24	0.89
female workers	(.05)	(.02)	(.06)	(.04)	(.57)	(.11)	(.15)	(.09)
female	0.28	0.79	0.30	0.69	1.71	0.83	0.33	0.90
unskilled workers	(.04)	(.02)	(.09)	(.06)	(.54)	(.11)	(.21)	(.11)
female skilled	2.15	1.63	1.14	2.00	15.37	1.46	0.14	2.35
workers	(.36)	(.10)	(.39)	(.18)	(4.93)	(.50)	(.57)	(.27)
male skilled	2.49	1.57	5.38	2.66	4.23	1.07	2.00	1.68
workers	(.25)	(.06)	(.56)	(.15)	(1.43)	(.19)	(.58)	(.19)
Wage-prod. gap for	0.41		0.48		-1.55		0.65	
female workers	0.52		0.39		-0.91		0.57	
female								
unskilled workers	-0.51		0.86		-13.91		1.21	
female skilled	-0.91		-2.73		-3.16		-0.32	
workers	-0.51		-2.13		-3.10		-0.32	
R-squared	0.12		0.16		0.17		0.17	
N	22,028		6,283		2,859		1,768	
		Machine Building Building Materials Timber			Chemical			
	Machine	e Building	Building	Materials	Tir	nber	Chei	mical
Wages and	Machine	e Building	Building	Materials	Tir	nber	Chei	mical
Wages and productivities	Machine	e Building	Building	Materials	Tir	nber	Chei	mical
								mical
productivities relative to	Log	Log	Log	Log	Log	Log	Log	
productivities relative to unskilled male	Log (Value		Log (Value		Log (Value		Log (Value	Log
productivities relative to	Log (Value Added)	Log (Wages)	Log (Value Added)	Log (Wages)	Log (Value Added)	Log	Log	Log (Wages)
productivities relative to unskilled male workers for	Log (Value Added) 0.69	Log (Wages)	Log (Value Added) 0.52	Log (Wages)	Log (Value Added)	Log (Wages)	Log (Value Added) 0.47	Log (Wages)
productivities relative to unskilled male workers for	Log (Value Added) 0.69 (.03)	Log (Wages) 0.86 (.01)	Log (Value Added) 0.52 (.06)	Log (Wages) 0.87 (.02)	Log (Value Added) 0.80 (.09)	Log (Wages) 0.94 (.03)	Log (Value Added) 0.47 (.03)	Log (Wages) 0.88 (.01)
productivities relative to unskilled male workers forfemale workersfemale	Log (Value Added) 0.69 (.03) 0.59	Log (Wages) 0.86 (.01) 0.83	Log (Value Added) 0.52 (.06) 0.45	Log (Wages) 0.87 (.02) 0.79	Log (Value Added) 0.80 (.09) 0.76	Log (Wages) 0.94 (.03) 0.94	Log (Value Added) 0.47 (.03) 0.34	Log (Wages) 0.88 (.01) 0.83
productivities relative to unskilled male workers forfemale workersfemale unskilled workers	Log (Value Added) 0.69 (.03) 0.59 (.03)	Log (Wages) 0.86 (.01) 0.83 (.01)	Log (Value Added) 0.52 (.06) 0.45 (.05)	Log (Wages) 0.87 (.02) 0.79 (.02)	Log (Value Added) 0.80 (.09) 0.76 (.09)	Log (Wages) 0.94 (.03) 0.94 (.03)	Log (Value Added) 0.47 (.03) 0.34 (.03)	Log (Wages) 0.88 (.01) 0.83 (.01)
productivities relative to unskilled male workers forfemale workersfemale unskilled workersfemale skilled	Log (Value Added) 0.69 (.03) 0.59 (.03) 2.98	Log (Wages) 0.86 (.01) 0.83 (.01) 1.71	Log (Value Added) 0.52 (.06) 0.45 (.05) 2.23	Log (Wages) 0.87 (.02) 0.79 (.02) 2.03	Log (Value Added) 0.80 (.09) 0.76 (.09) 2.91	Log (Wages) 0.94 (.03) 0.94 (.03) 1.49	Log (Value Added) 0.47 (.03) 0.34 (.03) 1.76	Log (Wages) 0.88 (.01) 0.83 (.01) 1.74
productivities relative to unskilled male workers forfemale workersfemale unskilled workersfemale skilled workers	Log (Value Added) 0.69 (.03) 0.59 (.03) 2.98 (.20)	Log (Wages) 0.86 (.01) 0.83 (.01) 1.71 (.05)	Log (Value Added) 0.52 (.06) 0.45 (.05) 2.23 (.51)	Log (Wages) 0.87 (.02) 0.79 (.02) 2.03 (.15)	Log (Value Added) 0.80 (.09) 0.76 (.09) 2.91 (.74)	Log (Wages) 0.94 (.03) 0.94 (.03) 1.49 (.19)	Log (Value Added) 0.47 (.03) 0.34 (.03) 1.76 (.22)	Log (Wages) 0.88 (.01) 0.83 (.01) 1.74 (.06)
productivities relative to unskilled male workers forfemale workersfemale unskilled workersfemale skilled workersmale skilled	Log (Value Added) 0.69 (.03) 0.59 (.03) 2.98 (.20) 3.74	Log (Wages) 0.86 (.01) 0.83 (.01) 1.71 (.05) 1.82	Log (Value Added) 0.52 (.06) 0.45 (.05) 2.23 (.51) 3.66	Log (Wages) 0.87 (.02) 0.79 (.02) 2.03	Log (Value Added) 0.80 (.09) 0.76 (.09) 2.91 (.74) 3.31	Log (Wages) 0.94 (.03) 0.94 (.03) 1.49 (.19) 1.69	Log (Value Added) 0.47 (.03) 0.34 (.03) 1.76 (.22) 3.14	Log (Wages) 0.88 (.01) 0.83 (.01) 1.74 (.06)
productivities relative to unskilled male workers for female workersfemale unskilled workersfemale skilled workersmale skilled workers	Log (Value Added) 0.69 (.03) 0.59 (.03) 2.98 (.20) 3.74 (.18)	Log (Wages) 0.86 (.01) 0.83 (.01) 1.71 (.05)	Log (Value Added) 0.52 (.06) 0.45 (.05) 2.23 (.51) 3.66 (.38)	Log (Wages) 0.87 (.02) 0.79 (.02) 2.03 (.15) 1.69	Log (Value Added) 0.80 (.09) 0.76 (.09) 2.91 (.74) 3.31 (.64)	Log (Wages) 0.94 (.03) 0.94 (.03) 1.49 (.19)	Log (Value Added) 0.47 (.03) 0.34 (.03) 1.76 (.22) 3.14 (.02)	Log (Wages) 0.88 (.01) 0.83 (.01) 1.74 (.06)
productivities relative to unskilled male workers forfemale workersfemale unskilled workersfemale skilled workersmale skilled workers Wage-prod. gap	Log (Value Added) 0.69 (.03) 0.59 (.03) 2.98 (.20) 3.74	Log (Wages) 0.86 (.01) 0.83 (.01) 1.71 (.05) 1.82	Log (Value Added) 0.52 (.06) 0.45 (.05) 2.23 (.51) 3.66	Log (Wages) 0.87 (.02) 0.79 (.02) 2.03 (.15) 1.69	Log (Value Added) 0.80 (.09) 0.76 (.09) 2.91 (.74) 3.31	Log (Wages) 0.94 (.03) 0.94 (.03) 1.49 (.19) 1.69	Log (Value Added) 0.47 (.03) 0.34 (.03) 1.76 (.22) 3.14	Log (Wages) 0.88 (.01) 0.83 (.01) 1.74 (.06)
productivities relative to unskilled male workers forfemale workersfemale unskilled workersfemale skilled workersmale skilled workers	Log (Value Added) 0.69 (.03) 0.59 (.03) 2.98 (.20) 3.74 (.18)	Log (Wages) 0.86 (.01) 0.83 (.01) 1.71 (.05) 1.82	Log (Value Added) 0.52 (.06) 0.45 (.05) 2.23 (.51) 3.66 (.38)	Log (Wages) 0.87 (.02) 0.79 (.02) 2.03 (.15) 1.69	Log (Value Added) 0.80 (.09) 0.76 (.09) 2.91 (.74) 3.31 (.64)	Log (Wages) 0.94 (.03) 0.94 (.03) 1.49 (.19) 1.69	Log (Value Added) 0.47 (.03) 0.34 (.03) 1.76 (.22) 3.14 (.02)	Log (Wages) 0.88 (.01) 0.83 (.01) 1.74 (.06)
productivities relative to unskilled male workers for female workersfemale unskilled workersfemale skilled workersmale skilled workers Wage-prod. gap for	Log (Value Added) 0.69 (.03) 0.59 (.03) 2.98 (.20) 3.74 (.18) 0.17	Log (Wages) 0.86 (.01) 0.83 (.01) 1.71 (.05) 1.82	Log (Value Added) 0.52 (.06) 0.45 (.05) 2.23 (.51) 3.66 (.38) 0.35	Log (Wages) 0.87 (.02) 0.79 (.02) 2.03 (.15) 1.69	Log (Value Added) 0.80 (.09) 0.76 (.09) 2.91 (.74) 3.31 (.64)	Log (Wages) 0.94 (.03) 0.94 (.03) 1.49 (.19) 1.69	Log (Value Added) 0.47 (.03) 0.34 (.03) 1.76 (.22) 3.14 (.02) 0.45	Log (Wages) 0.88 (.01) 0.83 (.01) 1.74 (.06) 1.71
productivities relative to unskilled male workers for female workersfemale unskilled workersfemale skilled workersmale skilled workers Wage-prod. gap forfemale workersfemale workersfemale skilled	Log (Value Added) 0.69 (.03) 0.59 (.03) 2.98 (.20) 3.74 (.18) 0.17	Log (Wages) 0.86 (.01) 0.83 (.01) 1.71 (.05) 1.82	Log (Value Added) 0.52 (.06) 0.45 (.05) 2.23 (.51) 3.66 (.38) 0.35	Log (Wages) 0.87 (.02) 0.79 (.02) 2.03 (.15) 1.69	Log (Value Added) 0.80 (.09) 0.76 (.09) 2.91 (.74) 3.31 (.64) 0.08	Log (Wages) 0.94 (.03) 0.94 (.03) 1.49 (.19) 1.69	Log (Value Added) 0.47 (.03) 0.34 (.03) 1.76 (.22) 3.14 (.02) 0.45	Log (Wages) 0.88 (.01) 0.83 (.01) 1.74 (.06)
productivities relative to unskilled male workers for female workersfemale unskilled workersfemale skilled workersmale skilled workers Wage-prod. gap forfemale workersfemale unskilled workersfemale	Log (Value Added) 0.69 (.03) 0.59 (.03) 2.98 (.20) 3.74 (.18) 0.17 0.24 -1.27	Log (Wages) 0.86 (.01) 0.83 (.01) 1.71 (.05) 1.82	Log (Value Added) 0.52 (.06) 0.45 (.05) 2.23 (.51) 3.66 (.38) 0.35 0.33 -0.20	Log (Wages) 0.87 (.02) 0.79 (.02) 2.03 (.15) 1.69	Log (Value Added) 0.80 (.09) 0.76 (.09) 2.91 (.74) 3.31 (.64) 0.08 0.18 -1.42	Log (Wages) 0.94 (.03) 0.94 (.03) 1.49 (.19) 1.69	Log (Value Added) 0.47 (.03) 0.34 (.03) 1.76 (.22) 3.14 (.02) 0.45 0.49 -0.02	Log (Wages) 0.88 (.01) 0.83 (.01) 1.74 (.06)
productivities relative to unskilled male workers for female workersfemale unskilled workersfemale skilled workersmale skilled workers Wage-prod. gap forfemale workersfemale workersfemale skilled	Log (Value Added) 0.69 (.03) 0.59 (.03) 2.98 (.20) 3.74 (.18) 0.17 0.24	Log (Wages) 0.86 (.01) 0.83 (.01) 1.71 (.05) 1.82	Log (Value Added) 0.52 (.06) 0.45 (.05) 2.23 (.51) 3.66 (.38) 0.35 0.33	Log (Wages) 0.87 (.02) 0.79 (.02) 2.03 (.15) 1.69	Log (Value Added) 0.80 (.09) 0.76 (.09) 2.91 (.74) 3.31 (.64) 0.08 0.18	Log (Wages) 0.94 (.03) 0.94 (.03) 1.49 (.19) 1.69	Log (Value Added) 0.47 (.03) 0.34 (.03) 1.76 (.22) 3.14 (.02) 0.45 0.49 -0.02	Log (Wages) 0.88 (.01) 0.83 (.01) 1.74 (.06)

Table B9 contd.

Joint production function and earnings equation estimates: Industrial categories

some production function an	1	-				
	Fo	ood	Textile		Tailoring	
Wages and productivities relative to unskilled male workers for	Log (Value Added)	Log (Wages)	Log (Value Added)	Log (Wages)	Log (Value Added)	Log (Wages)
	0.28	0.88	1.11	0.99	0.79	1.04
female workers	(.03)	(.02)	(.15)	(.04)	(.05)	(.03)
	0.30	0.84	0.50	0.94	0.81	1.00
female unskilled workers	(.03)	(.02)	(.04)	(.02)	(.07)	(.03)
	0.88	1.49	2.27	1.14	3.69	1.88
female skilled workers	(.22)	(.09)	(.37)	(.08)	(.50)	(.12)
	3.32	1.98	3.39	1.75	4.44	1.08
male skilled workers	(.31)	(.08)	(.57)	(.12)	(.84)	(.14)
Wage-productivity gap for						
female workers	0.53		-0.12		0.25	
female unskilled workers	0.54		0.44		0.19	
female skilled workers	0.61		-1.12		-1.80	
male skilled workers	-1.33		-1.65		-3.35	
R-squared	0.16		0.08		0.10	
N	15,095		17,328		9,323	
	_		Culture and Education		Leather	
	Pa	per			Lea	ather
Wages and productivities relative to unskilled male workers for	Log (Value	Log (Wages)	Log (Value		Log (Value	Log (Wages)
to unskilled male workers for	Log (Value Added) 0.66	Log (Wages)	Log (Value Added) 0.76	Log (Wages)	Log (Value Added) 0.67	Log (Wages)
	Log (Value Added) 0.66 (.05)	Log (Wages) 0.89 (.03)	Log (Value Added) 0.76 (.06)	Log (Wages) 0.88 (.02)	Log (Value Added) 0.67 (.06)	Log (Wages) 0.98 (.03)
to unskilled male workers forfemale workers	Log (Value Added) 0.66 (.05) 0.66	Log (Wages) 0.89 (.03) 0.94	Log (Value Added) 0.76 (.06) 0.72	Log (Wages) 0.88 (.02) 0.95	Log (Value Added) 0.67 (.06) 0.63	Log (Wages) 0.98 (.03) 1.01
to unskilled male workers for	Log (Value Added) 0.66 (.05) 0.66 (.06)	Log (Wages) 0.89 (.03) 0.94 (.03)	Log (Value Added) 0.76 (.06) 0.72 (.07)	Log (Wages) 0.88 (.02) 0.95 (.03)	Log (Value Added) 0.67 (.06) 0.63 (.07)	Log (Wages) 0.98 (.03) 1.01 (.04)
female workersfemale unskilled workers	Log (Value Added) 0.66 (.05) 0.66 (.06) 2.02	Log (Wages) 0.89 (.03) 0.94 (.03) 1.38	Log (Value Added) 0.76 (.06) 0.72 (.07) 3.12	Log (Wages) 0.88 (.02) 0.95 (.03)	Log (Value Added) 0.67 (.06) 0.63 (.07) 2.14	Log (Wages) 0.98 (.03) 1.01 (.04) 1.20
to unskilled male workers forfemale workers	Log (Value Added) 0.66 (.05) 0.66 (.06) 2.02 (.39)	Log (Wages) 0.89 (.03) 0.94 (.03) 1.38 (.13)	Log (Value Added) 0.76 (.06) 0.72 (.07) 3.12 (.61)	Log (Wages) 0.88 (.02) 0.95 (.03) 1.10 (.14)	Log (Value Added) 0.67 (.06) 0.63 (.07) 2.14 (.76)	Log (Wages) 0.98 (.03) 1.01 (.04) 1.20 (.22)
to unskilled male workers forfemale workersfemale unskilled workersfemale skilled workers	Log (Value Added) 0.66 (.05) 0.66 (.06) 2.02 (.39) 3.03	Log (Wages) 0.89 (.03) 0.94 (.03) 1.38 (.13)	Log (Value Added) 0.76 (.06) 0.72 (.07) 3.12 (.61)	Log (Wages) 0.88 (.02) 0.95 (.03) 1.10 (.14) 2.29	Log (Value Added) 0.67 (.06) 0.63 (.07) 2.14 (.76) 5.26	Log (Wages) 0.98 (.03) 1.01 (.04) 1.20 (.22) 1.83
to unskilled male workers forfemale workersfemale unskilled workersfemale skilled workersmale skilled workers	Log (Value Added) 0.66 (.05) 0.66 (.06) 2.02 (.39)	Log (Wages) 0.89 (.03) 0.94 (.03) 1.38 (.13)	Log (Value Added) 0.76 (.06) 0.72 (.07) 3.12 (.61)	Log (Wages) 0.88 (.02) 0.95 (.03) 1.10 (.14)	Log (Value Added) 0.67 (.06) 0.63 (.07) 2.14 (.76)	Log (Wages) 0.98 (.03) 1.01 (.04) 1.20 (.22)
to unskilled male workers forfemale workersfemale unskilled workersfemale skilled workersmale skilled workers Wage-productivity gap for	Log (Value Added) 0.66 (.05) 0.66 (.06) 2.02 (.39) 3.03 (.40)	Log (Wages) 0.89 (.03) 0.94 (.03) 1.38 (.13)	Log (Value Added) 0.76 (.06) 0.72 (.07) 3.12 (.61) 3.11 (.70)	Log (Wages) 0.88 (.02) 0.95 (.03) 1.10 (.14) 2.29	Log (Value Added) 0.67 (.06) 0.63 (.07) 2.14 (.76) 5.26 (1.14)	Log (Wages) 0.98 (.03) 1.01 (.04) 1.20 (.22) 1.83
to unskilled male workers forfemale workersfemale unskilled workersfemale skilled workersmale skilled workers Wage-productivity gap forfemale workers	Log (Value Added) 0.66 (.05) 0.66 (.06) 2.02 (.39) 3.03 (.40)	Log (Wages) 0.89 (.03) 0.94 (.03) 1.38 (.13)	Log (Value Added) 0.76 (.06) 0.72 (.07) 3.12 (.61) 3.11 (.70)	Log (Wages) 0.88 (.02) 0.95 (.03) 1.10 (.14) 2.29	Log (Value Added) 0.67 (.06) 0.63 (.07) 2.14 (.76) 5.26 (1.14)	Log (Wages) 0.98 (.03) 1.01 (.04) 1.20 (.22) 1.83
female workersfemale workersfemale unskilled workersfemale skilled workersmale skilled workers Wage-productivity gap forfemale workersfemale unskilled workers	Log (Value Added) 0.66 (.05) 0.66 (.06) 2.02 (.39) 3.03 (.40) 0.23 0.28	Log (Wages) 0.89 (.03) 0.94 (.03) 1.38 (.13)	Log (Value Added) 0.76 (.06) 0.72 (.07) 3.12 (.61) 3.11 (.70) 0.11	Log (Wages) 0.88 (.02) 0.95 (.03) 1.10 (.14) 2.29	Log (Value Added) 0.67 (.06) 0.63 (.07) 2.14 (.76) 5.26 (1.14) 0.41 0.38	Log (Wages) 0.98 (.03) 1.01 (.04) 1.20 (.22) 1.83
female workersfemale workersfemale unskilled workersfemale skilled workersmale skilled workers Wage-productivity gap forfemale workersfemale unskilled workersfemale unskilled workersfemale skilled workers	Log (Value Added) 0.66 (.05) 0.66 (.06) 2.02 (.39) 3.03 (.40) 0.23 0.28	Log (Wages) 0.89 (.03) 0.94 (.03) 1.38 (.13)	Log (Value Added) 0.76 (.06) 0.72 (.07) 3.12 (.61) 3.11 (.70) 0.11 0.23 -2.02	Log (Wages) 0.88 (.02) 0.95 (.03) 1.10 (.14) 2.29	Log (Value Added) 0.67 (.06) 0.63 (.07) 2.14 (.76) 5.26 (1.14) 0.41 0.38	Log (Wages) 0.98 (.03) 1.01 (.04) 1.20 (.22) 1.83
female workersfemale workersfemale unskilled workersfemale skilled workersmale skilled workers Wage-productivity gap forfemale workersfemale unskilled workersfemale skilled workersfemale skilled workers	Log (Value Added) 0.66 (.05) 0.66 (.06) 2.02 (.39) 3.03 (.40) 0.23 0.28 -0.64 -1.06	Log (Wages) 0.89 (.03) 0.94 (.03) 1.38 (.13)	Log (Value Added) 0.76 (.06) 0.72 (.07) 3.12 (.61) 3.11 (.70) 0.11 0.23 -2.02 -0.82	Log (Wages) 0.88 (.02) 0.95 (.03) 1.10 (.14) 2.29	Log (Value Added) 0.67 (.06) 0.63 (.07) 2.14 (.76) 5.26 (1.14) 0.41 0.38 -0.94 -3.43	Log (Wages) 0.98 (.03) 1.01 (.04) 1.20 (.22) 1.83
female workersfemale workersfemale unskilled workersfemale skilled workersmale skilled workers Wage-productivity gap forfemale workersfemale unskilled workersfemale unskilled workersfemale skilled workers	Log (Value Added) 0.66 (.05) 0.66 (.06) 2.02 (.39) 3.03 (.40) 0.23 0.28	Log (Wages) 0.89 (.03) 0.94 (.03) 1.38 (.13)	Log (Value Added) 0.76 (.06) 0.72 (.07) 3.12 (.61) 3.11 (.70) 0.11 0.23 -2.02	Log (Wages) 0.88 (.02) 0.95 (.03) 1.10 (.14) 2.29	Log (Value Added) 0.67 (.06) 0.63 (.07) 2.14 (.76) 5.26 (1.14) 0.41 0.38	Log (Wages) 0.98 (.03) 1.01 (.04) 1.20 (.22) 1.83

Table B10
Joint production function and earnings equation estimates: firms located in China's Special Economic Zones

	,	
Wages and productivities relative	Log	Log
to unskilled male workers for	(Value Added)	(Wages)
	0.79	0.01
female workers	(.05)	(.03)
	1.01	1.03
female unskilled workers	(.08)	(.03)
	3.97	2.51
female skilled workers	(.50)	(.12)
	7.01	2.99
male skilled workers	(.72)	(.13)
Wage-productivity gap for		
female workers	0.18	
female unskilled workers	0.03	
female skilled workers	-1.46	
male skilled workers	-4.00	
	0.20	
Log capital	(.01)	
	0.73	
Log labor	(.02)	
	0.04	0.04
Log R&D expenditure	(.00)	(.00)
	0.04	-0.08
Rural administrative subordination	(.03)	(.02)
R-squared	0.61	0.85
Correlation between equations	0.15	

WORKING PAPERS* Editor: Nils Gottfries

- 2008:16 Sören Blomquist and Håkan Selin, Hourly Wage Rate and Taxable Labor Income Responsiveness to Changes in Marginal Tax Rates. 31 pp.
- 2008:17 Jie Chen and Aiyong Zhu, The relationship between housing investment and economic growth in China: A panel analysis using quarterly provincial data. 26pp.
- 2009:1 Per Engström, Patrik Hesselius and Bertil Holmlund, Vacancy Referrals, Job
 Search, and the Duration of Unemployment: A Randomized Experiment.
 25 pp.
- 2009:2 Chuan-Zhong Li and Gunnar Isacsson, Valuing urban accessibility and air quality in Sweden: A regional welfare analysis. 24pp.
- 2009:3 Luca Micheletto, Optimal nonlinear redistributive taxation and public good provision in an economy with Veblen effects. 26 pp.
- 2009:4 Håkan Selin, The Rise in Female Employment and the Role of Tax Incentives. An Empirical Analysis of the Swedish Individual Tax Reform of 1971. 38 pp.
- 2009:5 Lars M. Johansson and Jan Pettersson, Tied Aid, Trade-Facilitating Aid or Trade-Diverting Aid? 47pp.
- 2009:6 Håkan Selin, Marginal tax rates and tax-favoured pension savings of the self-employed Evidence from Sweden. 32pp.
- 2009:7 Tobias Lindhe and Jan Södersten, Dividend taxation, share repurchases and the equity trap. 27pp.
- 2009:8 Che-Yuan Liang, Nonparametric Structural Estimation of Labor Supply in the Presence of Censoring. 48pp.
- 2009:9 Bertil Holmlund, Incentives in Business and Academia. 12pp.
- 2009:10 Jakob Winstrand, The Effects of a Refinery on Property Values The Case of Sweden. 27pp.
- 2009:11 Ranjula Bali Swain and Adel Varghese, The Impact of Skill Development and Human Capital Training on Self Help Groups. 28pp.
- 2009:12 Mikael Elinder. Correcting Mistakes: Cognitive Dissonance and Political Attitudes in Sweden and the United States. 25 pp.

* A list of papers in this series from earlier years will be sent on request by the department.

-

- 2009:13 Sören Blomquist, Vidar Christiansen and Luca Micheletto: Public Provision of Private Goods and Nondistortionary Marginal Tax Rates: Some further Results. 41pp.
- 2009:14 Mattias Nordin, The effect of information on voting behavior. 34pp.
- 2009:15 Anders Klevmarken, Olle Grünewald and Henrik Allansson, A new consumer price index that incorporates housing. 27 pp.
- 2009:16 Heléne L. Nilsson, How Local are Local Governments? Heterogeneous Effects of Intergovernmental Grants. 41pp.
- 2009:17 Olof Åslund, Per-Anders Edin, Peter Fredriksson and Hans Grönqvist, Peers, neighborhoods and immigrant student achievement evidence from a placement policy. 27 pp.
- 2009:18 Yunus Aksoy, Henrique S. Basso and Javier Coto-Martinez, Lending Relationships and Monetary Policy. 42 pp.
- 2009:19 Johan Söderberg, Non-uniform staggered prices and output persistence. 38 pp.
- 2010:1 Jonathan Gemus, College Achievement and Earnings. 43 pp.
- 2010:2 Susanne Ek and Bertil Holmlund, Family Job Search, Wage Bargaining, and Optimal Unemployment Insurance. 30 pp.
- 2010:3 Sören Blomquist and Laurent Simula, Marginal Deadweight Loss when the Income Tax is Nonlinear. 21 pp.
- 2010:4 Niklas Bengtsson, The marginal propensity to earn, consume and save out of unearned income in South Africa. 34 pp.
- 2010:5 Marcus Eliason and Henry Ohlsson, Timing of death and the repeal of the Swedish inheritance tax. 29 pp.
- 2010:6 Teodora Borota, Innovation and Imitation in a Model of North-South Trade. 44 pp.
- 2010:7 Cristiana Benedetti Fasil and Teodora Borota, World Trade Patterns and Prices: The Role of Productivity and Quality Heterogeneity. 24 pp.
- 2010:8 Johanna Rickne, Gender, Wages and Social Security in China's Industrial Sector. 48 pp.

See also working papers published by the Office of Labour Market Policy Evaluation http://www.ifau.se/ ISSN 1653-6975